



-96

TECHNICAL BASIS FOR DERIVING SEDIMENT QUALITY CRITERIA FOR NONIONIC ORGANIC CONTAMINANTS FOR THE PROTECTION OF BENTHIC ORGANISMS BY USING EQUILIBRIUM PARTITIONING

(U.S.) ENVIRONMENTAL PROTECTION AGENCY, WASHINGTON, DC

SEP 93



? 2. Report Date ? 3. Report Type And Dates Covered 33 ย 1 ° SEP 93 Ð SEP 93 t1 25 11 pp. 137030 บ์ล์ล • 5. Funding Numbers 9 4. Title And Subtitle 17 11 \mathbf{v} n TECHNICAL BASIS FOR DERIVING SEDIMENT QUALITY 0 t) ø **v** CRITERIA FOR NONIONIC ORGANIC CONTAMINANTS FOR 12 Ð " 6. Author(s) 12 t1 12 Ø Ð, ø "7. Performing Organization Name(s) And Address(es) °8. Performing Organization® 0 " US EPA ŧ. Report Number " OFFICE OF WATER e) 401 M ST., S.W. 12 " WASHINGTON, DC 20460 \mathbf{v} n 02 ^u 9. Sponsoring/Monitoring Agency Name(s) 10. Sponsoring/Monitoring^v Ω And Address(es) 0 Agency Report Number n O US EPA 0 822/R-93-011 \mathbf{n} O OFFICE OF WATER \mathbf{c} 401 M ST., S.W. 51 ø Ω " WASHINGTON, DC 20460 12 ² 11. Supplementary Notes 11 a c Ð u " 12a. Distribution/Availability Statement ° 12b. Distribution Code 12 0 Ð n 0 r. n 11 n 15 1 13. Abstract THIS REPORT PRESENTS THE TECHNICAL BASIS FOR ESTABLISHING C " SEDIMENT QUALITY CRITERIA FOR NONIONIC ORGANIC CHEMICALS, USING EQUILIBRIUM 12 11 PARTITIONING (EqP). EqP IS CHOSEN BECAUSE IT ADDRESSES THE TWO PRINCIPAL ŧ٢ TECHNICAL ISSUES THAT MUST BE RESOLVED: THE VARYING JIOAVAILABILITY OF 11 I n Ð CHEMICALS IN SEDIMENTS AND THE CHOICE OF THE APPROPRIATE BIOLOGICAL EFFECTS **8** : CONCENTRATION. 0 n Ð Ð n (7 a 12 Ø t : " 14. Subject Terms SEDIMENT QUALITY CRITERIA, • 15. Number Of Pages 12 n BENTHIC ORGANISMS, EQUILIBRIUM PARTITIONING, 68 11 ¹³ NONIONIC ORGANIC CHEMICALS Ũááááááááááááááááááááááááááááá ti ° 16. Price Code 11 ÚÁRÁRIZÁREZERE A CONTRACTION CONTRACTION CONTRACTION CONTRACTION CONTRACTION CONTRACTION CONTRACTION CONTRACTION " 17. Report Security " 18. Page Security " 19. Abstract Security • 20. Media " o 12 Classication o Classication Classication o 63 0 21 2 ti.

United States Environmental Protection Agency

Cilice of Water (WH-553) Washington, DC 20460 EPA-822-R-93-011 September 1993

PB95-137030



Technical Basis for Deriving Sediment Quality Criteria for Nonionic Organic Contaminants for the Protection of Benthic Organisms by Using Equilibrium Partitioning

17. Punted on Recycled Paper

U.S. OPARTINES IN FORMULA STREET U.S. OPARTINES IN FORMULA STREET Princed Technology Prince Spacephile, Vegalak State

This document presents the technical basis EPA has used in establishing the proposed methodology for deriving sediment quality criteria for the protection of benthic organisms from nonionic organic chemicals. It was issued in support of EPA regulations and policy initiatives involving the application of biological and chemical assessment techniques to control toxic pollution to surface waters and sediments. This document does not establish or affect legal rights or obligations. It does not establish a binding norm and is not finally determinative of the issues addressed. Agency decisions in any particular case will be made applying the law and regulalions on the basis of specific facts when permits are issued or regulations promulgated. This document is expected to be revised periodically to reflect advances in this rapidly evolving area.

This report has been reviewed by the Health and Ecological Criteria Division, Office of Science and Technology, U.S. Environmental Protection Agency, as well as other pertinent and interested offices in the Agency, and approved for publication. Mention of trade names or commercial products does not constitute endorsement or recommendations for use.

CONTENTS

ABSTRACT	1
OVERVIEW	3
Toxicity and Bioavailability of Chemicals in Sediments	. 3
Partitioning of Nonionic Organic Chemicals	. 5
Effects Concentration	, 7
BACKGROUND	7
Rationale for Selecting the EoP Method	. 8
Relationship to WOC Methodology	. 8
Applications of SOC	8
TOXICITY AND BIOAVAILABILITY OF CHEMICALS IN SEDIMENTS	9
Toxicity Experiments	. 9
Bioaccumulation	12
Conclusion	12
SORPTION OF NONIONIC ORGANIC CHEMICALS	14
Partitioning in Particle Suspensions	14
Particle concentration effect	14
Organic carbon fraction	16
Dissolved Organic Carbon (DOC) Complexing	17
Phase Distribution in Sediments	17
Bioavailability of DOC Complexed Chemicals	18
Field Observations of Partitioning in Sediments	19
Organic carbon normalization	19
Sediment/pore water partitioning	23
Laboratory toxicity tests	24
Organic Carbon Normalization of Biological Responses	26
Toxicity and bioaccumulation experiments	26
Bioaccumulation and organic carbon normalization	28
Determination of the Route of Exposure	31
ADDITION THE AD WAR AS THE FERENCE FUELS FOR RENTHING ADDANISMS	94
Method Polating A sub Sociality	37
	34
Comparison of the Sensitivity of Benthic and Water Column Species	32
	32
Au species	34

1

ív

Calastics	Criter	1a (wq	ĮC) C	or	lCe	eni	tra	tio	n	5 \	/er	su	S													
Colonizatio	on Exp	erin	ner	its	٠	٠	٠		٠	٠	•	٦	٦	•	٠	•	٠	•	•	•	٠	•	٠	٠	٠	•	٠	36
Conclusions .		•	• •	•	•	٠	•	٠	•	·	٠	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	••	37
ENERATION OF S	QC .					,	•			•				,	,			,			•							37
Parameter Va	ues .	•														•												37
Measurement	of Kow		• •																									38
Literature	Kow .			•																				ļ				40
Estimated	Kow .		• •	•																								40
Kow select	ion .															•									·			42
Koc detern	ninatio	n						•												,			÷				Ż	42
Species Sensit	ivitv .											_		Ċ										Ċ			•	43
Quantification	of Un	cer	tair	ntv	Å	550	- Ci	iat	ed	w	i+ł	د د	ŝ	Ċ	•	•	•	•	•	·		•	•	•	•	·	•	44
Minimum Rea	mirem	ent	e tr	, 		101		0 Q	ŝ	~ ~			~	-	•	•	•	•	•	•	ſ	•	•	•	•	•	•	-14 16
Laborator	v octan	പപ	wai	lor.	na	•p• irti	itiz	 	~~` ~~	∽ ∖ofi	Fir	io.	nt	·	•	•	•	•	•	•	•	·	•	·	·	•	•	40
Final chro	nic vali				P				÷			16	***	•	•	•	·	•	•	•	•	•••	·	•	•	•	•	40
Sediment	tovicits	ite:	et i		•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	47
Analytical	ntoced	tun.		•	•	•	•	•	•	•	•	•	•	•	•	•	•	٠	•	•	•	•	•		•	•	•	47
Conclusio	n	~~~~	-0,			•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	·	40 40
Evample Calc	ulation	•	• •	•	•	•	•	٠	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	40
Eigld Data	uration	ð	• •	•	•	•	•	•	•	•	•	•	•	•	•	٠	•	•	٠	٠	٠	•	•	٠	•	٠	•	98
STOPET A	• • •	•	• •	•	*	•	•	•	•	•	•	•	٠	•	٠	•	•	•	•	•	·	•	•	·	٠	•	•	49
SIOKEI G	dia .	سم	••••	سرا		مەرد	•	-				•	•	·	•	•	•	٠	·	•	•	·	٠	٠	•	•	•	50
Correct E	iatus a	na	Ire	na.	51	.т с	vg,	rai	no	aa	a	٠	•	•	•	•	•	÷	٠	•	٠	٠	•	٠	•	•	•	50
Corps of b	nginee	15 0	246	d .	•	-	•	•	•	•	•	•	٠	•	•	٠	•	•	•	٠	•	•	٠	•	٠	٠	•	51
NCLUSIONS .																												55
Research Nee	de la							-		,			-	•		-	•	•			•	•		•	·	•		

Abstract

The purpose of this report is to present the technical basis for establishing sediment quality criteria for nonionic organic chemicals, using equilibrium partitioning (EqP). Equilibrium partitioning is chosen because it addresses the two principal technical issues that must be resolved: the varying bioavailability of chemicals in sediments and the choice of the appropriate biological effects concentration.

The data that are used to examine the question of varying bioavailability across sediments are from toxicity and bioaccumulation experiments using the same chemical and test organism but different sediments. It has been found that if the different sediments in each experiment are compared, there is essentially no relationship between sediment chemical concentrations on a dry weight basis and biological effects. However, if the chemical concentrations in the pore wate: of the sediment are used (for chemicals that are not highly hydrophobic) or if the sediment chemical concentrations on an organic carbon basis are used, then the biological effects occur at similar concentrations (typically within a factor of two) for the different sediments. Most importantly, the effects concentrations are the same as, or they can be predicted from, the effects concentration determined in water-only exposures.

The EqP methodology rationalizes these results by assuming that the partitioning of the chemical between sediment-organic carbon and pore water is at equilibrium. In each of these phases, the fugacity or activity of the chemical is the same at equilibrium. As a consequence, it is assumed that the organism receives an equivalent exposure from a water only-exposure or from any equilibrated phase: either from pore water via respiration: or from sediment carbon, via ingestion; or from a mixture of the routes. Thus, the pathway of exposure is not significant. The biological effect is produced by the chemical activity of the single phase or the equilibrated system.

Sediment quality criteria (SQC) for nonionic organic chemicals are based on the chemical concentration in sediment organic carbon. For highly hydrophobic chemicals this is necessary because the pore water concentration is, for those chemicals, no longer a good estimate of the chemical activity. The pore water concentration is the sum of the free chemical activity, and the concentration of chemical activity, and the concentration of chemical complexed to dissolved organic carbon, which is not bioavailable. Using the chemical concentration in sediment organic carbon eliminates this ambiguity.

2

SQC also require that a chemical concentration be chosen that is sufficiently protective of benthic organisms. The final chronic value (FCV) from the U.S. Environmental Protection Agency (EPA) water quality criteria is proposed. An analysis of the data compiled in the water quality criteria documents demonstrates that benthic species, defined as either epibenthic or infaunal species, have a similar sensitivity to water column species. This similarity is the case if the most sensitive species are compared and if all species are compared. The results of benthic colonization experiments also support the use of the FCV. Thus, if effects concentrations in sediments can be accurately predicted using the K_{0c} and data from water-only tests, the SQC protecting benthic species can be predicted using the K_{0c} and FCV.

Equilibrium partitioning cannot remove all the variation in the experimentally observed sediment-effects concentration and the concentration predicted from water-only exposures. A variation factor of approximately four to five remains. Thus, a quantification of this uncertainty should accompany the SQC.

The derivation of SQC requires that a minimum database be available. This includes: (1) the octanol/water partition coefficient of the chemical, which should be measured with modern experimental techniques, which appear to remove the large variation in reported values, (2) the derivation of the final chronic value, which should also be updated to include the most recent toxicological information, and (3) an SQC check test to establish variation of the Eq? prediction. The SQC is then the FCV x K_{oc} with confidence limits based on SQC check tests.

OVERVIEW

This report presents the technical basis for establishing sediment quality criteria (5QC) for nonionic organic chemicals using the equilibrium partitioning (EqP) method. The term sediment quality criteria, as used herein, refers to numerical concentrations for individual chemicals that are applicable across the range of sediments encountered in practice. Sediment quality criteria are intended to be predictive of biological effects. As a consequence, they can be used in much the same way as final chronic values (FCV) are used in water quality criteria—as the concentration of a chemical that is protective of benthic aquatic life.

The specific regulatory uses of SQC have not been established. However, the range of potential applications is quite large because the need for the evaluation of potentially contaminated sediments arises in many contexts. SQC are meant to be used with direct toxicity testing of sediments as a method of evaluation. They provide a chemical by chemical specification of what sediment concentrations are protective of benthic aquatic life.

This overview (Section 1) summarizes the evidence and the major lines of reasoning of the EqP methodology, with supporting references cited in the body of the report. Section 2 reviews the background that led to the need for SQC and also the selection of the EqP methodology. Section 3 reviews the development of concentration-response curves for porewater concentrations and sediment organic-carbon normalized concentrations to determine toxicity and bioavailability in contaminated sediments. It also presents analyses of sediment toxicity and bioaccumulation experiments. Section 4 reviews the partitioning of nonionic organic chemicals to sediments using laboratory and field studies. Section 5 reviews a comparison of benthic and water column species using aquatic toxicity data contained in EPA's Water Quality Criteria (WQC) documents to show the applicability of WQC as the effects levels for benthic organisms. Section 6 reviews the computation of an SQC and presents an analysis for quantifying the uncertainty associated with SQC. This section also presents minimum data requirements and example calculations and compares the SQC computed for five chemicals to field data. Section 7 presents conclusions and further research needs. Section 8 lists the references used in this document.

Toxicity and Bioavailability of Chemicals in Sediments

Establishing SQC requires a determination of the extent of the bioavailability of sediment associated chemicals. It has frequently been observed that similar concentrations of a chemical, in units of mass of chemical per mass of sediment dry weight (e.g., micrograms chemical per gram sediment $(\mu g/g)$ can exhibit a range in toxicity in different sediments. If the purpose of SQC is to establish chemical concentrations that apply to sediments of differing types, it is essential that the reasons for this varying bioavailability be understood and explicitly included in the criteria. Otherwise the criteria cannot be presumed to be applicable across sediments of differing properties.

The importance of this issue cannot be overemphasized. For example, if 1 µg/g of Kepone is the LCso for an organism in one sediment and 35 µg/g is the LC50 in another sediment, then unless the cause of this difference can be associated with some explicit sediment properties it is not possible to decide what would be the LC50 of a third sediment without performing a toxicity test. The results of toxicity tests used to establish the toxicity of chemicals in sediments would not be generalizable to other sediments. Imagine the situation if the results of toxicity tests in water depended strongly on the particular water source, for example, Lake Superior versus well water. Until the source of the differences was understood, it would be fruitless to attempt to establish WQC. For this reason, bioavailability is a principal focus of this report.

The key insight into the problem of quantifying the bioavailability of chemicals in sediments was that the concentration-response curve for the biological effect of concern can be correlated not to the total sediment-chemical concentration (micrograms chemical per gram sediment), but to the interstitual water or pore water concentration (micrograms chemical per liter pore water). In addition, the effects concentration found for the pore water is essentially equal to that found in water-only exposures. Organism mortality, growth rate, and bioaccumulation data are used to demonstrate this correlation, which is a critical part of the logic behind the EqP approach to developing SQC. For nonionic organic chemicals, the concentration-response curves correlate equally well with the sediment-chemical concentration on a sediment-organic carbon basis.

These observations can be rationalized by assuming that the pore water and sediment carbon are in equilibrium and that the concentrations are related by a partition coefficient, Koc, as shown in Figure 1. The name equilibrium partitioning (EqP) describes this assumption. The rationalization for the equality of water-only and sediment-exposure-effects concentrations on a pore water basis is that the sediment-pore water equilibrium system (Fig. 1, right) provides the same exposure as a water-only exposure (Fig. 1, left). The chemical activity is the same in each system at equilibrium. It should be pointed out that the EqP assumptions are only approximately true; therefore, predictions from the model have an inherent uncertainty. The data presented below illustrate the degree to which EqP can rationalize the observations.



Figure 1.—Diagram of the organism exposure routes for a water-only exposure (left and a sediment exposure (right). Equilibrium partitioning refers to the assumption that an equilibrium exists between the chemical sorbed to the particulate sediment organic carbon and the pore water. Kee is the organic carbon partition coefficient.

Figure 2 presents mortality data for various chemicals and sediments compared to pore water concentrations when normalized on a toxic unit basis. Pore water toxic units are the ratio of the measured pore water concentration to the LCso from water-only toxicity tests. Three different sediments are tested for each chemical as indicated. The EqP model predicts that the pore water LCso will equal the water-only LCso which is obtained from a separate water-only exposure toxicity test. Define:

pore water toxic unit

(1) (pore water concentration) (1) (water-only LC50)

Therefore, a toxic unit of one occurs when the pore water concentration equals the water-only LC 50, at which point it would be predicted that 30 percent mortality would be observed. The correlation of observed mortality to predicted pore water toxic units in Figure 2 demonstrates (a) the efficacy of using pore water concentrations to remove sediment-tosediment differences and (b) the applicability of the water-only effects concentration and, by implication, the validity of the EqP model. By contrast, the mortality versus sediment chemical concentration on a dry weight basis varies dramatically from sediment to sediment. This will be presented subsequently. The equality of the effects concentration on a pore water basis suggests that the route of exposure is via pore water. However, the equality of the effects concentration on a sediment-organic carbon basis, which is demonstrated below, suggests that the ingestion of sediment-organic carbon is the primary route of exposure. It is important to realize that if the sediment and pore water are in equilibrium, then the effective exposure concentration is the same regardless of exposure route. Therefore, it is not possible to determine the primary route of exposure from equilibrated experiments.

Whatever the route of exposure, the correlation to pore water suggests that if it were possible to either measure the pore water chemical concentration, or predict it from the total sediment concentration and the relevant sediment properties such as the sediment organic carbon concentration, then that concentration could be used to quantify the exposure concentration for an organism. Thus, the partitioning of chemicals between the solid and the liquid phase in a sediment becomes a necessary component for establishing SQC.

In addition, if it were true that benthic organisms are as sensitive as water column organisms and the evidence to be presented appears to support this supposition—then SQC could be established using the FCV from WQC documents as the effects concentration for benthic organisms. The apparent equality between the effects concentration as meas-



Figure 2.—Mortality versus predicted pore water toxic units for five chemicals and three sediments per chemical, Sediment types are indicated by the single hatching (lowest organic carbon content), cross-hatching (intermediate organic carbon content), and filled symbols (highest organic carbon content). See Tables 1 and 2 for data sources. Predicted pore water toxic units are the ratio of the pore water concentration to the water-only LCso (Eqn. 1).

ured in pore water and in water-only exposures (Fig. 2) supports using an effects concentration derived from water-only exposures.

The calculation procedure for establishing SQC is as follows. If FCV $(\mu g/L)$ is the final chronic WQC for the chemical of interest, then the SQC $(\mu g/g \text{ sediment})$ are computed using the partition coefficient K_p (L/kg sediment) defined as the ratio of chemical concentration in the sediment and in the pore water at equilibrium.

$$SQC = K_p FCV \times 0.001 \frac{kg}{g}$$
(2)

This is the fundamental equation from which SQC are generated. Its utility depends on the existence of a methodology for quantifying partition coefficients.

Partitioning of Nonionic Organic Chemicals

The partitioning of nonionic organic chemicals to soil and sediment particles is reasonably well understood, and a standard model exists for describing the process. The hydrophobicity of the chemical is quantified by using the octanol/water partition coefficient, Kow. The sorption capacity of the sediment is determined by the mass fraction of organic carbon for the sediment, foc. For sediments with foce 0.2 percent by weight, the organic carbon appears to be the predominant phase for chemical sorption. The partition coefficient, K_p , the ratio of sediment concentration, C_5 , to pore water concentration, C_d , is given by

$$K_p = \frac{C_s}{C_d} = f_{oc} K_{oc}$$
(3)

where K_{0c} is the partition coefficient for sediment organic carbon.

The only other environmental variable that has a dramatic effect on partitioning appears to be the particle concentration in the suspension in which K_p is measured. There is considerable controversy regarding the mechanism responsible for the particle concentration effect, and a number of explanations have been offered. However, all the interpretations yield the same result for sediment/pore water partitioning, namely that $K_{oc} \neq K_{ow}$ for sediments.

Using Equations 2 and 3, a SQC is calculated from

$$SQC = f_{oc}K_{oc}FCV.$$
 (4)

This equation is linear in the organic carbon fraction, for As a consequence, the relationship can be expressed as

$$\frac{SQC}{f_{oc}} = K_{oc}FCV.$$
 (5)

It we define

ť,

$$SQC_{oc} = \frac{SQC}{for}$$
(6)

as the organic carbon-normalized SQC concentration (microgram chemical per gram organic carbon), then

$$SQC_{oc} = K_{oc}FCV.$$
 (7)

Thus, we arrive at the following important conclusion: For a specific chemical having a specific K_{oc} , the organic carbon-normalized sediment concentration, SQC_{oc}, is independent of sediment properties.

Hydrophobic chemicals also tend to partition to colloidal-sized organic carbon particles that are commonly referred to as dissolved organic carbon, or DOC. Although DOC affects the apparent pore water concentrations of highly hydrophobic chemicals, the DOC-bound fraction of the chemical appears not to be bioavailable and Equation 7 for SQCoc still applies. Therefore, we expect that toxicity in sediment can be predicted from the water-only effects concentration and the K_{0c} of the chemical. The utility of these ideas can be tested with the same mortality data as those in Figure 2 but restricted to nonionic organic chemicals for which organic carbon normalization applies. The concept of sediment toxic units is useful in this regard. These units are computed as the ratio of the organic carbon-normalized sediment concentrations, C_{1}/f_{0c} , and the predicted sediment LC50 using Koc and the water-only LC50. That is,

predicted	C ₂₄	(8)
sediment		
toxic unit	Noc (Water-only LC50)	

Figure 3 presents the percent mortality versus predicted sediment toxic units. The correlation is similar to that obtained using the pore water concentrations in Figure 2. The cadmium data are not included because its partitioning is not determined by

.



Organic Carbon Normalization

Figure 3.—Mortality versus predicted sediment toxic units. Predicted sediment toxic units are the ratio of the organic carbonnormalized sediment chemical concentration to the predicted sediment LC50 (Eqn. 8). Sediment types are indicated by the single hatching (lowest organic carbon content), cross-hatching (intermediate organic carbon content), and filled symbols (highest organic carbon content). See Tables 1 and 2 for date sources. K_{oc} values are computed from K_{ow} for DDT (5.84), endrin (4.84), fluoranthene (5.00), dieldrin (5.25), phenanthrene (4.46), and acenaphthene (3.76) with Equation 11. K_{ow} for DDT is the log average of the reported values in the Log P database [75]. The kepone K_{oc} is the log mean of the ratio of organic carbon-normalized kepone concentration to pore water-kepone concentration from the toxicity data set. K_{ow} s for the remaining compounds were computed by the U.S. EPA, Environmental Research Laboratory, Athens, Georgia. Methods are presented later in this document.

sediment organic carbon. The predicted sediment toxic units for each chemical follow a similar concentration-response curve independent of sediment type. The data demonstrate that 50 percent mortality occurs at about one sediment toxic unit, independent of chemical, species of organism, or sediment type, as expected if the EqP assumptions are correct.

If the assumptions of EqP were exactly true, and there were no experimental variability or measurement error, then all data in Figures 2 and 3 should predict 50 percent mortality at one toxic unit. There is an uncertainty factor of approximately four to five in the results. This variation reflects inherent variability in these experiments and phenomena that have not been accounted for in the EqP model. It also appears to be the limit of the accuracy and precision that can be expected.

Effects Concentration

The development of SQC requires an effects concentration for benthic organisms. Because many of the organisms used to establish the WQC are benthic, perhaps the WQC are adequate estimates of the effects concentrations for benthic organisms. To examine this possibility, the acute toxicity database, which is used to establish the WQC, is segregated into benthic and water column species, and the relative sensitivities of each group are compared. Figure 4 compares the acute values for the most sensitive ben-



Figure 4.—A comparison of the final acute values (FAV) for water column versus benthic organisms. Each data point represents an FAV for a particular chemical in either a freshwater or a saltwater exposure. The data are from the WQC or draft criteria documents. See Table 4 for data sources.

thic (epibenthic and infaunal) species to the mostsensitive water column species. The data are from the 40 freshwater and 30 saltwater U.S. Environmental Protection Agency (EPA) criteria documents. Although there is considerable scatter, these results, a more detailed analysis of all the acute toxicity data, and the results of benthic colonization experiments support the contention of equal sensitivity.

BACKGROUND

Under the Clean Water Act (CWA), the EPA is responsible for protecting the chemical, physical, and biological integrity of the nation's waters. In keeping with this responsibility, EPA published WQC in 1980 for 64 of the 65 priority pollutants and pollutant categories listed as toxic in the CWA. Additional water quality documents that update criteria for selected consent decree and new chemicals have been published since 1980. These WQC are numerical concentration limits that are protective of human health and aquatic life. Although these criteria play an important role in assuring a healthy aquatic environment, they are not sufficient to ensure appropriate levels of environmental and human health protection.

Toxic contaminants in bottom sediments of the nation's lakes, rivers, wetlands, and coastal waters create the potential for continued environmental degradation even where water column contaminant levels comply with established WQC. The absence of defensible SQC makes it difficult to assess the extent of sediment contamination, implement measures to limit or prevent additional contamination from occurring, or to identify and implement appropriate remediation as needed.

As a result of the need to assist regulatory agencies in making decisions concerning contaminated sediment, the EPA's Office of Science and Technology, Health and Ecological Criteria Division, established a research team to review alternative approaches to assess sediment contamination. Sediment contamination and related problems were the subject of a conference [1]. Alternative approaches to establishing SQC [2] and their merits and deficiencies were discussed [3]. Additional efforts were undertaken to identify the scope of national sediment contamination [4] and to review proposed approaches for addressing contaminated sediments [5, 6]. The EqP method was selected because it provides the most practical, scientifically defensible, and effective regulatory tool for addressing individual nonionic chemicals associated with contaminated sediments on a national basis [7].

Rationale for Selecting the EqP Method

The principal reasons for the selection of the EqP method include the following:

- The EqP method was most likely to yield sediment criteria that are predictive of biological effects in the field and defensible when used in a regulatory context. These criteria address the issue of bioavailability and are founded on the extensive biological effects database used to establish national WQC.
- Sediment criteria can be readily incorporated into existing regulatory operations because a unique numerical sediment-specific criterion can be established for any chemical and compared to field measurements to assess the likelihood of significant adverse effects.
- Sediment criteria provide a simple and cost-effective means of screening sediment measurements to identify areas of concern and provide information to regulators in a short period of time.
- The method takes advantage of the data and expertise that led to the development of national WQC.
- The methodology can be used as a regulatory tool to ensure that uncontaminated sites are protected from attaining unacceptable levels of contamination.

Relationship to WQC Methodology

The first question to be answered is this: Why not use the WQC procedure for the development of SQC? A detailed methodology has been developed that presents the supporting logic, establishes the required minimum toxicological data set, and specifies the numerical procedures to be used to calculate the criteria values [8]. Further, WQC developed through this methodology are routinely used in the regulation of effluent discharges. Therefore, it is only natural to extend these methods directly to sediments.

The WQC are based on total chemical concentration, so the transition to using dissolved chemical concentration for those chemicals that partition to a significant extent would not be difficult. The experience with site-specific modifications of the national WQC has demonstrated that the water-effect ratio, the ratio of chemical concentrations in site water to laboratory water that produces the same effect, has averaged 3.5 [9, 10]. The implication is that differences of this magnitude result from variations in site-specific water chemistry and are not an overwhelming impediment to nationally applicable numerical WQC.

The WQC are based on using the total chemical concentration as a measure of bioavailable chemical concentration. However, the use of total sediment chemical concentration as a measure of the bioavailable-or even potentially bioavailable-chemical concentration is not supported by the available data [11]. The results of recent experiments indicate that sediments can differ in toxicity by factors of 100 or more for the same total chemical concentration. This difference is a significant obstacle. Without a quantitative estimate of the bioavailable chemical concentration in a sediment, it is impossible to predict a sediment's toxicity on the basis of chemical measurements, regardless of the method used to assess biological impact-be it laboratory toxicity experiments or field data sets comprising benthic biological and chemical sampling [12-15].

Without a unique relationship between chemical measurements and biological end points that applies across the range of sediment properties that affect bioavailability, the cause and effect linkage is not supportable. If the same total chemical concentration is 100 times more toxic in one sediment than it is in another, how can we set universal SQC that depend only on the total sediment chemical concentration? Any SQC that are based on total sediment concentration have a potential uncertainty of this magnitude. Thus, bioavailability must be explicitly considered for any sediment evaluation methodology that depends on chemical measurements to establish defensible SQC.

Applications of SQC

SQC that are reasonably accurate in their ability to predict the potential for biological impacts are useful in many activities [16]. Sediment quality criteria can play a significant role in the identification, monitoring, and cleanup of contaminated sediment sites on a national basis and provide a basis to ensure that sites that are uncontaminated will remain so. In some cases, sediment criteria alone are sufficient to identify and establish cleanup levels for contaminated sediments. In other cases, they must be supplemented with biological sampling and testing before decisions are made.

In many ways, sediment criteria developed using the EqP methodology are similar to WQC. However, their application may be quite different. In most cases, contaminants exceeding WQC in the water column need only be controlled at the source to eliminate unacceptable adverse impacts. Contaminated sediments have often been in place for quite some time, and controlling the source of that pollution (if the source still exists) may not be sufficient to alleviate the problem. The difficulty is compounded because the safe removal and treatment or disposal of contaminated sediments can be laborious and expensive.

Sediment criteria can be used as a means for predicting or identifying the degree and extent of contaminated areas such that more informed regulatory decisions can be made. Sediment criteria will be particularly valuable in monitoring applications in which sediment contaminant concentrations are gradually approaching the criteria. The comparison of field measurements to sediment criteria will provide reliable warning of potential problems. Such an early warning provides an opportunity to take corrective action before adverse impacts occur.

TOXICITY AND BIOAVAILABILITY OF CHEMICALS IN SEDIMENTS

A key insight into the problem of quantifying the bioavailability of chemicals in sediments was that the concentration-response curve for the biological effect of concern could be correlated, not to the total sediment chemical concentration (micrograms chemical per gram dry sediment) but to the pore water concentration (micrograms chemical per liter pore water) [17]. However, these results do not necessarily imply that pore water is the primary route of exposure because all exposure pathways are at equal chemical activity in an equilibrium experiment (see Fig. 1) and the route of exposure cannot be determined. Nevertheless, this observation was the critical first step in understanding bioavailability of chemicals in sediments.

Toxicity Experiments

A substantial amount of data has been assembled that addresses the relationship between toxicity and pore water chemical concentrations. Table 1 lists the sources and characteristics of these experiments. Some of these data are presented in Figures 5 to 8. The remaining data are presented elewhere in this document. In Figures 5 to 8 the biological response-mortality or growth rate suppression—is plotted versus the total sediment concentration in the top panel, and versus the measured pore water concentration in the bottom panel. Table 2 summarizes the LC50 and EC50 estimates and 95 percent confidence limits for these data on a total sediment and pore water basis, as well as the water-only values.

The results from kepone experiments (Fig. 5) are illustrative of the general trends in these data [17, 18]. For the low organic carbon sediment ($f_{oc} = 0.09$ percent), the 50th percentile total kepone concentra-

CHEMICAL	ORGANISM	SEDIMENT SOURCE	EXPOSURE DURATION (DAYS)	BIOLOGICAL END POINT	REFERENCE	FIGURE
ACENAPHTHENE	Eohaustonus estuenus	South Beach, OR McKinney Stough, OR Eckman Stough, OR	10	Mortality	i (56)	23
ACENAPHTHENE	Leptocherius plumulosus	South Beach, OR McKinney Strugh, OR Eckman Stough, OR	10	Mortainty	(56)	23
CYPERMETURIN	Churonomus tentans	River and pond	1	Body burden	(22)	8
ODT	Hysiella attece	Soap Creek, Mercer Lake	10	Mortality	[20.21]	7
DIELORIN	Chironomus tentans	Airport Pond. MN	10	Mortainty	(55)	26
DIELDRIN	Hysielia azteca	West Bearskin Lake, MN Pequaywan Lake, MN	10	Mortality	[53]	
DIELDRIN	Hydiolla Alleca	Auport Pond, MN	10	Mortality	[54]	24
ENDRIN	Diporeia sp.	Lake Michigan	10	Mortality	(52)	23
ENDRIN	Hyaialla Bzteca	Soap Creek, Mercer Lake	10	Mortality	[20.21]	7
FLUORANTHENE	Rhepoxynius abionius	Amended One Beach, OR	10	Mortality	[57]	23
FLUORANTHENE	Rhepoxymius abronius	Yaquina Bay, OR	10	Mortainy	[19]	6
KEPONE	Chironomus tentans	Soil	14	Body burden	[17.24]	
KEPONE	Chironomus tentans	Soil	14	Growth	[17]	5
KEPONE	Chiranonius tentans	Soil	14	Mortality	[17]	5
PERMETHRIN	Chuonomus tentans	River and pond	1	i Body burden	[22]	
PHENANTHRENE	Ephauslonus estuanus	South Beach, OR McKinney Stouth, OR Eckman Stough, OR	10	Montahty	(56)	23
PHENANTHRENE	Ceptocherivs plumulosus	South Beach, OR McKinney Slough, OR Eckmen Stough, OR	10	Mortality	(56)	23







Dry Weight Normalization



•

LCEO AND ECGO CHEMICAL (END POINT) TOTAL SEDIMENT (HE/E) Ac (%) PORE WATER (ME/L) ORGANIC CARBON (HE/S) WATER ONLY (ME/L) AFFERENCE 0.90 0.90 (0.73-1.10) 29.9 (25.3-35.6) 1.000 (811-1.220) 26.4 (22.7-30.6) [17] KEPONE 1,50 6.9 (5.85-8.12) 31.3 (25.7-38.1) 460 (390-541) (MORTALITY) 12.0 35.2 (30.6-40.5) 18.6 (15.7-21.9) 293 (255-337) 0.09 0.46 (0.42-0.51) 17.1 (15.7-18.7) 511 (467-567) 16.2 (15.0-17.5) (17) KEPONE (GROWTH) 1.5 9.93 (7.74-12.8) 48.5 (34.6-67.8) 662 (516-1.050) 12.0 37.3 (31.5-44.2) 20.1 (16.7-24.1) 311 (262-368) 0.2 21.9 (19.6-24.4) 3.2 (2.85-3.59) 1.600 (1.430-1.800) [19] FLUORANTHENE 30.9 (27.0-35.4) 2.130 (1.850-2.420) 0.3 6.4 (5.56-7.27) (MORTALITY) 0.5 10.7 (8.34-13.7) 22.2 (17.5-29.3) 2.140 (1.670-2.740) 10.3 (8.74-12.2) 0.74 (0.67-0.82) 344 (291-405) 3.0 0.45 (0.38-0.53) [20] 100 7.2 17.5 (12.5-24.3) 1.45 (1.20-1.75) 243 (174-338) 0.48 (0.42-0.55) (MORTALITY) 10.5 44.9 (36.7-55.0; 0.77 (0.67-0.89) 428 (350-524) 0.52 (0.45-0.60) 3.0 1.54 (1.18-2.00) 51.3 (39.3-66.7 [21] DDT (MORTALITY) 3.0 4.16 (3.91-4.42) 139 (130-147) 11.0 10.95 (9.34-12.9) 99.6 (84.9-117) 3.0 3.39 (2.61-4.41) 1.80 (1.44-2.24) 113 (87.0147) 4.81 (4.46-5.20) [20] ENDRIN 1.92 (1.55-2.36) 83.1 (66.4-104) 6.1 5.07 (4.05-6.36) 3.39 (3.10-4.98) (MORTALITY) 11.2 1.74 (1.37-2 20) 5.91 (4.73-7.37) 52.8 (42.2-65.8) 3.71 (3.11-4.44) 3.0 4.76 (3.70-8.13) 2.26 (1.67-3.05) 159 (123-204) 1211 ENDRIN 11.0 18,9 (13.6-26.3) 3.75 (2.72-5,19) 172 (124-239) (MORTALITY) 11.0 10.5 (8.29-12.7) 2.81 (2.44-3.23) 95.8 (75.4-115)

The LCsos and ECsos and the 95% confidence limits in parentheses are computed by the modified Spearman-Karber method [123].

tion for both Chironomus tentans mortality (LC₅₀) and growth rate reduction from a life cycle test (EC₅₀) are <1 μ g/g. By contrast, the 1.5 percent organic carbon sediment EC₅₀ and LC₅₀ are approximately 7 and 10 μ g/g, respectively. The high organic carbon sediment (12 percent) exhibits still higher LC₅₀ and EC₅₀ values on a total sediment kepone concentration basis (35 and 37 μ g/g, respectively).

However, as shown in the bottom panels, essentially all the mortality data collapse into a single curve and the variation in growth rate is significantly reduced when the pore water concentrations are used as the correlating concentrations. On a pore water basis, the biological responses as measured by LC50 or EC50 vary by approximately a factor of two, whereas when they are evaluated on a total sediment kepone basis, they exhibit an almost 40-fold range in kepone toxicity.

The comparison between the pore water effects concentrations and the water-only results indicates that they are similar. The pore water LC505 are 19 to 30 μ g/L, and the water-only exposure LC50 is 26 μ g/L. The pore water EC505 are 17 to 49 μ g/L, and the water-only EC50 is 16 μ g/L (Table 2).

Laboratory experiments have also been performed to characterize the toxicity of fluoranthene [19] to the sediment-dwelling marine amphipod *Rhepoxynius abrotius*. Figure 6 presents the *R. abrotius* mortality data for the fluorenthene experiment. The results of the fluoranthene experiments parallel those for kepone. The sediment with the lowest organic carbon content (0.2 percent) exhibits the lowest LC50 on a total sediment concentration basis (3.2 $\mu g/g$) and as the organic carbon concentration increases (0.3 and 0.5 percent) the LC50 increases (6.4 and 10.7 $\mu g/g$). On a pore water basis, the data again collapse to a single concentration-response curve and the LC505 differ by less than 50 percent.

Figure 7 presents mortality data for DDT and endrin using the (reshwater amphipod Hyalella azteca [20, 21]. The responses for DDT [20] are similar to those observed for kepone, cadmium, and fluoranthene. On a total sediment concentration basis the organism responses differ for the various sediments (LCsos are 10.3 to 45 µg/L), but on a pore water basis the responses are again similar (LCsos are 0.74 to 1.4 µg/L) and comparable to the water-only LCsos of approximately 0.5 µg/L. The DDT data reported by Schuytema et al. [21] are more variable. By contrast,



Dry Weight Normalization

Figure 7.—Comparison of percent mortality of H. atteca to DDT (left) and endrin (nght) concentrations in bulk sediment (top) and pore water (bottom) for sediments with varying organic cation concentrations [20, 21].

the organism survival for endrin exposures on a dry weight basis varies by a factor of almost six among the six sediments. The LC50s are 3.4 to 18.9 μ g/g. The pore water LC50s were less variable, 1.7 to 3.8 μ g/L and comparable to the water-only exposure LC50 of approximately 4 μ g/L (Table 2).

Bloaccumulation

12

A direct measure of chemical bioavailability is the amount of chemical retained in organism tissues. Hence, tissue bioaccumulation data can be used to examine the extent of chemical bioavailability. Chironomus tentans was exposed to two synthetic pyrethroids, cypermethrin and permethrin, spiked into three sediments, one of which was laboratory-grade sand [22]. The bioaccumulation from the sand was approximately an order of magnitude higher than it was from the organic carbon-containing sediments for both cypermethrin and permethrin (Fig. 8, top panels).

On a pore water basis, the bioaccumulation appears to be approximately linear and independent of sediment type (bottom panels). The mean bioaccumulation factor (BAF) for cypermethrin (and permethrin) varies from 6.2 to 0.6 (4.0 to 0.23) (μ g/g organism/ μ g/g sediment) as sediment f_{oc} increases (Table 3). By contrast the mean BAFs on a pore water basis vary by less than a factor of two.

Bioaccumulation was also measured by Adams et al. [17, 23] and Adams [24] in the C. tentans-kepone experiments presented previously (Figure 3). The body burden variation on a total sediment basis is over two orders of magnitude (BAF = 600 to 3.3 µg/g organism/µg/g sediment), whereas the pore water bioaccumulation factor is within a factor of four (5,200 to 17,600 µg/kg organism/µg/L), with the very low organic carbon sediment exhibiting the largest deviation (Table 3).

Conclusion

These observations—that organism concentration response and bioaccumulation from different sediments can be reduced to one curve if pore water is considered as the concentration that quantifies exposure—can be interpreted in a number of ways. However, these results do not necessarily imply that pore water is the primary route of exposure because all exposure pathways are at equal chemical activity in an



13



Figure 8.-Comparison of C. tentans body burden of cypermethrin (left) and permethrin (right) versus concentration in bulk sediment (top) and pore water (bottom) for sediments with varying organic carbon concentrations [22].

Pore Water Con

entration (vg/L)

		BIOACCUMULA	TION FACTORS	
CHEMICAL	foc (%)	TOTAL SEDIMENT	PORE WATER WE'NE ORGANISM WE'L	REFERENCE
	<0.1	6.21 (4.41-8.01)	80.1 (73.5-86.7)	
CVBER LITURIN	2.3	0.50 (0.30-0.71)	51.3 (43.8-58.8)	(22)
GTPERMEINHIN	3.7	0.60 (0.37-0.83)	92.9 (87.0-98.8)	
	<0.1	4.04 (2.89-5.20)	39.7 (25.0-54.3)	
0704074074	2.3	0.38 (0.17-0.59)	52.5 (22.6-82.4)	[22]
	3.7	0.23 (0.18-0.28)	29.7 (15,6-43.7)	
	0.09	600 (308-692)	17.600 (6,540-28.600)	
VEDOUE	1.50	20 (4.8-35.2)	5.180 (1.970-8.390)	[17,23.24]
	12	3.3 (0.3-6.3)	5.790 (2.890-8.702)	

Pore Water Concentration (ug/L)

*95% confidence limits shown in parentheses.

19757

equilibrium experiment. The route of exposure cannot be determined, as we can see by comparing the concentration-response correlations to pore water and organic carbon-normalized sediment concentrations. That both are equally successful at correlating the data suggests that neither the pore water nor the sediment exposure pathway can be implicated as the primary exposure route.

In order to relate pore water exposure to sedituent carbon exposure, it is necessary to establish the relationship between these two concentrations. Therefore, an examination of the state of the art of predicting the partitioning of chemicals between the solid and the liquid phase is required. This examination is described in the following section.

SORPTION OF NONIONIC ORGANIC CHEMICALS

Partitioning in Particle Suspensions

A number of empirical models have been suggested to explain the sorption of nonionic hydrophobic organic chemicals to natural soils and sediment particles (see Karickhoff [25] for an excellent review). The chemical property that indexes hydrophobicity is the octanol/water partition coefficient, Kow. The important particle property is the weight fraction of organic carbon, f_{oc} . Another important environmental variable appears to be the particle concentration itself.

In many experiments using particle suspensions, the partition coefficients have been observed to decrease as the particle concentration used in the experiment is increased [26]. Very few experiments have been done on settled or undisturbed sediments; therefore, the correct interpretation of particle suspension experiments is of critical importance. It is not uncommon for the partition coefficient to decrease by two to three orders of magnitude at high particle concentrations. If this partitioning behavior is characteristic of bedded sediments, then quite low partition coefficients would be appropriate, which would result in lower sediment chemical concentrations for SQC. If, however, this phenomenon is an artifact or a result of a phenomenon that does not apply to bedded sediments, then a quite different partition coefficient would be used. The practical importance of this issue requires a detailed discussion of the particle concentration effect.

Particle concentration effect. For the reversible (or readily desorbable) component of sorption, a particle interaction model (PIM) has been proposed that accounts for the particle concentration effect and predicts the partition coefficient of nonionic hydrophobic chemicals over a range of nearly seven orders of magnitude with a log10 prediction standard error of 0.38 [27]. The reversible component partition coefficient, K_p^{μ} , is the ratio of reversibly bound chemical concentration, C_s ($\mu g/kg$ dry weight), to the dissolved chemical concentration, C_d ($\mu g/L$):

 $C_{s} = K_{p}^{s} C_{d}.$ (9)

The PIM model for $K \stackrel{\bullet}{P}$, is

$$K_{p}^{*} = \frac{f_{oc}K_{oc}}{1 + mf_{oc}K_{oc}/v_{r}}$$
(10)

where

 K_{p}^{*} = reversible component partition coefficient (L/kg dry weight)

- Koc = particle organic carbon partition coefficient (L/kg organic carbon)
- foc = particle organic carbon weight fraction (kg organic carbon/kg dry weight)
- *n* = particle concentration in the suspension (kg dry weight/L)
- $v_x = 1.4$, an empirical constant (unitless).

The regression of K_{oc} to the octanol/water coefficient, K_{ow} , yields

$$\log_{10}K_{oc} = 0.00028 + 0.983 \log_{10}K_{ow}$$
 (11)

which is that essentially Koc approximately equals Kow. Figure 9 presents the observed versus predicted reversible component partition coefficients using this model [27]. A substantial fraction of the data in the regression is at high particle concentrations (mfocKow > 10), where the partitioning is determined only by the solids concentration and vx. The low particle concentration data (m/ocKow < 1) are presented in Figure 10 for the conventional adsorption (left) and reversible component (right) partition coefficient, Kp, normalized by f_{oc} , that is $K_{oc} = K_p / f_{oc}$. The relationship Koc = Kow is demonstrated from the agreement between the line of perfect equality and the data. It is important to note that while Equation 10 applies only to the reversible component partition coefficient, Kp, the equation: Kp = for Kow applies to the conventional adsorption partition coefficient as well (Fig. 10, left).

A number of explanations have been offered for the particle concentration effect. The most popular is the existence of an additional third sorbing phase or complexing component that is associated

19758



Figure 9.—Comparison of observed reversible component partition coefficient to calculated partition coefficient using Equation 10 [27].



Partition Coefficient - m foc Kow < 1

Figure 10.—Comparison of the adsorption (left) and reversible component (right) organic carbon-normalized partition coefficient. Kor, to the octanol/water partition coefficient, Kor, for experiments with low solids concentrations: $m_{loc}K_{ow}<1$. The line represents equality [27].

with the particles but is inadvertently measured as part of the dissolved chemical concentration because of experimental limitations. Colloidal particles that remain in solution after particle separation [28, 29] and dissolved ligands or macromolecules that desorb from the particles and remain in solution [30-33] have been suggested. It has also been suggested that increasing particle concentration increases the degree of particle aggregation, decreasing the surface area and hence the partition coefficient [34]. The effect has also been attributed to kinetic effects [25].

16

Sorption by nonseparated particles or complexing by dissolved organic carbon can produce an apparent decrease in partition coefficient with increasing particle concentration if the operational method of measuring dissolved chemical concentration does not properly discriminate the truly dissolved or free chemical concentration from the complexed or colloidally sorbed portion. However, the question is not whether improperly measured dissolved concentrations can lead to an apparent decrease in partition coefficient with increasing particle concentrations. The question is whether these third-phase models explain all (or most) of the observed partition coefficient-particle concentration relationships.

An alternate possibility is that the particle concentration effect is a distinct phenomena that is a ubiquitous feature of aqueous-phase particle sorption. A number of experiments have been designed explicitly to exclude possible third-phase interferences. Particle concentration effects are displayed in the resuspension experiment for polychlorinated biphenyls (PCBs) and metals [35-37], in which particles are resuspended into a reduced volume of supernatant, and in the dilution experiment in which the particle suspension is diluted with supernatant from a parallel vessel [35]. It is difficult to see how thirdphase models can account for these results because the concentration of the colloidal particles is constant while the concentration of the sediment particles varies substantially.

The model (Eqn. 10) is based on the hypothesis that particle concentration effects result from an additional desorption reaction induced by particleparticle interactions [27]. It has been suggested that actual particle collisions are responsible [38]. This interpretation relates v_{x} to the collision efficiency for desorption and demonstrates that it is independent of the chemical and particle properties, a fact that has been experimentally observed [27, 36].

It is not necessary to decide which of these mechanisms is responsible for the effect if all the possible interpretations yield the same result for sediment/pore water partitioning. Particle interaction models would predict that $K_{0c} \approx K_{0w}$ because the particles are stationary in sediments. Third-phase models would also relate the free (i.e., uncomplexed) dissolved chemical concentration to particulate concentration via the same equation. As for kinetic effects, the equilibrium concentration is again given by the relationship $K_{oc} = K_{ow}$. Thus there is unanimity on the proper partition coefficient to be used in order to relate the free dissolved chemical concentration to the sediment concentration: $K_{oc} = K_{ow}$.

Organic carbon fraction. The unifying parameter that permits the development of SQC for nonionic hydrophobic organic chemicals that are applicable to a broad range of sediment types is the organic carbon content of the sediments. This development can be shown as follows: The sediment/pore water partition coefficient, K_p , is given by

$$K_{p} = f_{oc} K_{oc} \approx f_{oc} K_{ow}$$
(12)

and the solid phase concentration is given by

$$C_{s} = f_{oc} K_{oc} C_{d}$$
(13)

where C₅ is the concentration on sediment particles. An important observation can be made that leads to the idea of organic carbon-normalization. Equation 12 indicates that the partition coefficient for any nonionic organic chemical is linear in the organic carbon fraction, foc. The partitioning data examined in Figure 10 can be used to examine the linearity of Kp to foc. Figure 11 compares Kp / Kow to foc for both the adsorption and the reversible component partition coefficients. The data are restricted to mfor Kow < 1 to suppress particle effects. The line indicates the expected linear relationship in Equation 12. These data and an analysis presented below appear to support the linearity of partitioning to a value of for = 0.2 percent. This result and the toxicity experiments examined below suggest that for for > 0.2 percent, organic carbon-normalization is valid.

As a consequence of the linear relationship of C_s and f_{oc} , the relationship between sediment concentration, C_s , and free dissolved concentration, C_d , can be expressed as

$$\frac{C_s}{f_{oc}} = K_{oc}C_d.$$
(14)

If we define

$$C_{s,oc} = \frac{C_s}{f_{oc}}$$
(15)

as the organic carbon normalized sediment concentration (ug chemical/kg organic carbon), then from Equation 14:

$$C_{s,oc} = K_{oc}C_{d}.$$
 (16)

Partition Coefficient - m foc Kow < 1 Adsorption **Reversible Component** Log10 Kp/Kow Log10 Kp/Ko -7 č DD1 0.01 1.00 100.00 0.01 0,10 10.00 0.10 1.00 100,00 .10.00 foc (%) toc (%)

Figure 11.—Comparison of the normalized partition coefficients for adsorption (left) and reversible component sorption (right) to sediment organic carbon. The data are restricted so that particle effects are not expected to be significant; $ml_{oc}K_{ow}$ < 1. The line represents perfect agreement [27].

Therefore, for a specific chemical with a specific Koc, the organic carbon-normalized total sediment concentration, $C_{s,oc}$, is proportional to the dissolved free concentration, C_d , for any sediment with $f_{oc} > 0.2$ percent. This latter qualification is judged to be necessary because at $f_{oc} < 0.2$ percent the other factors that influence partitioning (e.g., particle size and sorption to nonorganic mineral fractions) become relatively more important [25]. Using the proportional relationship given by Equation 16, the concentration of free dissolved chemical can be predicted from the normalized sediment concentration and Koc. The free concentration is of concern as it is the form that is bioavailable. The evidence is discussed in the next section.

Dissolved Organic Carbon (DOC) Complexing

In addition to partitioning to particulate organic carbon (POC) associated with sediment particles, hydrophobic chemicals can also partition to the organic carbon in colloidal sized particles. Because these particles are too small to be removed by conventional filtration or centrifugation they are operationally defined as DOC. Because sediment interstitial waters frequently contain significant levels of DOC, it must be considered in evaluating the phase distribution of chemicals.

A distinction is made between the free dissolved chemical concentration, C_d , and the DOC-

complexed chemical, C_{DOC} . The partition coefficient for DOC, K_{DOC} , is analogous to K_{oc} as it quantifies the ratio of DOC-bound chemical, C_{DOC} , to the free dissolved concentration, C_d :

$$C_{DOC} = m_{DOC}K_{DOC}C_d$$
 (17)

17

where m_{DOC} is the DOC concentration. The magnitude of K_{DOC} and the DOC concentration determine the extent of DOC complexation that takes place. Thus, it is important to have estimates of these quantities when calculating the level of free dissolved chemicals in sediment pore waters.

A recent compilation of K_{DOC} together with additional experimental determinations is available [39]. A summary that compares the partitioning of six chemicals to POC, natural DOC, and Aldrich humic acid (HA) is shown on Figure 12. The magnitude of the partition coefficients follow the order: POC > HA > natural DOC. The upper bound on K_{DOC} would appear to be $K_{DOC} = K_{oc}$, the POC partition coefficient.

Phase Distribution In Sediments

Chemicals in sediments are partitioned into three phases: free chemical; chemical sorbed to POC, and chemical sorbed to DOC. To evaluate the partitioning among these three phases, consider the mass balance for total concentration C_{T} :

 $C_T = \Theta C_d + m \int_{OC} K_{OC} C_d + \Theta m DOC K_{DOC} C_d$ (18)



18

Figure 12.—Partition coefficients of chemicals to particulate organic carbon (POC). Aldrich humic acid, and natural OOC. Benzo(ajpyrene (BaP); 2.2',4,4'.5.5' hexachlorobiphenyl (HCBP); DDT; 2.2',5.5' letrachlorobiphenyl (TCBP); pyrege (PYR); 4-monochlorobiphenyl (MCBP). Data from Eadle et al. (39).

where α is the sediment porosity (volume of water/volume of water plus solids) and *m* is the sediment solids concentration (mass of solids/volume of water plus solids). The three terms on the right side of the equation are the concentration of free chemical in the interstitial water, and that sorbed to the POC and DOC, respectively. Hence, from Equation 18 the free dissolved concentration can be expressed as

$$C_{\rm d} = \frac{C_{\rm T}}{\varrho + m f_{\rm oc} K_{\rm OS} + \varrho m_{\rm DOC} K_{\rm DOC}},$$
(19)

The concentration associated with the particle carbon (Eqn. 16) and DOC (Eqn. 17) can then be calculated. The total pore water concentration is the sum of the free and DOC complexed chemical, so that

 $C_{\text{pore}} = C_d + C_{\text{DOC}} = C_d(1 + m_{\text{DOC}}K_{\text{DOC}}). \quad (20)$

Figure 13 illustrates the phase partitioning behavior of a system for a unit concentration of a chemical with the following properties: $K_{oc} = K_{DCC} = 10^{6} L/kg$, $f_{oc} = 2.0$ percent, m = 0.5 kg solids/L sediment, and m_{DOC} varies from 0 to 50 mg/L, a reasonable range for pore waters [40]. With no DOC present, the pore water concentration equals the free concentration. As DOC increases, the pore water concentration increases because of the increase in complexed chemical, C_{DOC} . Accompanying this increase in C_{OC} is a slight—in fact, insignificant—decrease in C_{d} (Eqn. 19) and a proportional decrease in C_{s} (Eqn. 16).



Figure 13.—Partition coefficients of chemicals to particulate organic carbon (POC). Aldrich humic acid, and natural DOC. Benzol alpyrene (BaP); 2,2',4,4',5,5' hexachtorobiphenyl (HCBP); DDT; 2,2',5,5'-tetrachtorobiphenyl (TCBP); pyrene (PYR): 4-monochtorobiphenyl (MCBP). Data from Eadle et al. (39).

It is important to realize that the free chemical concentration, C_d, can be estimated directly from $C_{1,\infty}$, the organic carbon-normalized sediment concentration, using Equation 16, and that the estimate is independent of the DOC concentration. However, to estimate C_d from the pore water concentration requires that the DOC concentration and K_{DOC} be known. The assumption C_{pore} = C_d is clearly not warranted for very hydrophobic chemicals. For these cases C_{hoc} gives a more direct estimate of the free dissolved bioavailable concentration, C_d, than does the pore water concentration.

Bioavailability of DOC-Complexed Chemicals

The proportion of a chemical in pore water that is complexed to DOC can be substantial (Fig. 13). Hence, the question of bioavailability of DOC-complexed chemical can be important in assessing toxicity directly from measured pore water concentraions. A significant quantity of data indicates that DOC-complexed chemical is not bioavailable. Fish [41] and amphipod [42] uptake of polycyclic aromatic hydrocarbons (PAHs) are significantly reduced by adding DOC. An example is shown in Figure 14 for a freshwater amphipod [42]. For a highly hydrophobic chemical such as benzolalpyrene (BaP) the effect is substantial, whereas for less hydrophobic chemicals (e.g., phenanthrene) the reduction in uptake rate is insignificant. This result was expected because, for a fixed amount of DOC, the quantity of DOC-complexed chemical decreases with decreasing KDOC (Eqn. 17).



Figure 14.—Average uptake rate of chemicals by Pontoporeia hoyi with (filled) and without (hatched) DOC present. Benzo[a]pyrene (BaP): 2.2',4,4'tetrachiorobiphenyl (TCBP); pyrene, phonanthrene. Data from Landrum et al. [42].

The quantitative demonstration that DOCcomplexed chemicals are not bioavailable requires an independent determination of the concentration of complexed chemical. Landrum et al. [42] have developed a Cia reverse-phase HPLC column technique that separates the complexed and free chemical. Thus it is possible to compare the measured DOC-complexed chemical to the quantity of complexed chemical inferred from the uptake experiments, assuming that all the complexed chemical is not bioavailable [42, 43]. As shown on Figure 15, although the KDOC inferred from uptake suppression is larger than that inferred from the reverse-phase separation for HA, these data support the assumption that the DOCcomplexed fraction, CDOC, is not bioavailable. Hence the bioavailable form of dissolved chemical is Cd. the free uncomplexed component. This is an important observation because it is Ca that is in equilibrium with Cstor, the organic carbon-normalized sediment concentration (Eqn. 15).

Field Observations of Partitioning in Sediments

An enormous quantity of laboratory data exists for partitioning in particle suspensions. However, pore water and sediment data from field samples are scarce. Two types of data from field samples are ex-





19

Figure 15.—Comparison of the DOC partition coefficient calculated from the suppression of chemical uptake versus the C18 reversed-phase HPLC column estimate. Circles are Aldrich humic acid; triangles are interstitial water DOC. Chemicals are listed in Figure 14 caption (also enthracene and benzo(a)anthracene).

amined. The first is a direct test of the partitioning equation $C_{s,oc} = K_{oc} C_d$, which is independent of the DOC concentration. The second examines the sediment and pore water concentrations and accounts for the DOC that is present.

Organic carbon normalization. Consider a sediment sample that is segregated into vanous size classes after collection. The particles in each class were in contact with the pore water. If sorption equilibrium has been attained for each class, then, letting $C_s(j)$ be the particle chemical concentration of the *j*th size class, it is true that

$$C_{s}(j) = f_{oc}(j) K_{oc}C_{d}$$
(21)

where $f_{oc}(j)$ is the organic carbon fraction for each size class j. On an organic carbon-normalized basis this equation becomes

$$C_{s,oc}(j) = K_{oc}C_{d} \qquad (22)$$

where $C_{s.oc}(j) = C_s(j) / f_{oc}(j)$. This result indicates that the organic carbon-normalized sediment concentration of a chemical should be equal in each size class because K_{oc} and C_d are the same for each size class. Thus a direct test of the validity of both organic carbon normalization and EqP would be to examine whether $C_{s.oc}(j)$ is constant across size classes in a sediment sample.

Data from three field studies, Prahl [44], Evans et al. [45], and Delbeke et al. [46], can be used to test this prediction. In Prahl's study, sediment cores were collected at three stations near the Washington State coast (Stations 4, 5 and 7). These were sieved into a silt-and-clay sized fraction (<64 μ m), and a sand-sized fraction (>64 μ m). This latter fraction was further separated into a low density fraction (<1.9 g/cm3) and the remaining higher density sandsized particles. The concentrations of 13 individual PAHs were measured in each size fraction.

It is important to realize that these size fractions are not pure clay, silt, or sand but are natural particles in the size classes denoted by clay, silt, and sand. The organic carbon fractions, shown on Figure 16, range from 0.2 percent for the high-density sandsized fraction to greater than 30 percent for the lowdensity fraction. This exceeds two orders of magnitude and essentially spans the range usually found in practice. For example, 90 percent of the estuarine and coastal sediments sampled for the National Status and Trends program exceed 0.2 percent organic carbon [47].



Organic Carbon Fractions

Figure 16.—The organic carbon fractions (% dry weight) in the low-density fraction .64 μ m, <1.9 g/cm²; the sandsized fraction >64 μ m, >1.9 g/cm³; the silt/clay-sized fraction <64 μ m. Numbered stations as indicated. Data from [44].

Figure 17 (top) compares the dry weight-normalized clay/silt sized fraction sediment PAH concentrations. $C_s(j)$, to the sand-sized high- and low-density PAH concentrations on a dry weight basis. The dry weight-normalized data have distinctly different concentrations—the low-density high-organic carbon fraction is highly enriched, whereas the sand-sized fraction is substantially below the clay/silt fraction concentrations. Figure 17 (bottom) presents the same data but on an organic carbon-normalized basis, $C_{\text{s.ecc}}(j)$. In contrast to dry weight normalization, the PAH concentrations are essentially the same in each size class, as predicted by Equation 22.

In the field data from Evans et al. [45] sediments were collected at five sites along the River Derwent, Derbyshire. United Kingdom, and separated into six sediment size classes. The size classes were representative of clay and silt (<63 μ m) to course sand (1.0 to 2.0 mm). Organic carbon content and total PAH were measured in each sediment size class. Figure 18 presents the size classes and associated organic carbon c.ntent. Evans et al. attribute the bimodal distribution of f_{0c} to two types of organic matter. Organic matter in the 1.0 to 2.0 mm size class may be from fragmentary plant material while the size classes less then 500 μ m organic carbon content is the result of aging humic material. The organic content in this study ranges from 2.0 to 40 percent.

Figure 19 presents a comparison of PAH concentration for different sediment classes for dry weight normalization and organic carbon normalization. The top left panel compares PAH concentrations on sand (63 μ m - 500 μ m) and clay/silt (< 63 μ m) on a dry weight basis. The top right panel compares PAH concentrations on course sand (0.5 µm - 2.0 µm) and clav/silt (< 63 µm) on a dry weight basis. The data indicates that the PAH concentration is higher in the course sand fraction of sediment. Recall from Figure 18 that the clay/silt and course sand fractions contain higher fraction organic carbon content. The bottom panels of Figure 19 present the organic carbon normalized comparison of PAH concentrations by sediment class. For both panels, the organic carbon normalized PAH concentrations are similar regardless of the sediment size class as predicted by Equation 22

Lastly, Delbeke et al. [46] collected sediments from seven sites in the Belgian continental shelf and the Scheldt estuary. These sites were analyzed for eight PCB congeners and organic carbon in the bulk sediment and clay/silt (< 63 μ m) sediment fraction. In addition, analyses of the samples were done to determine the percent of size fractions ranging from 300 μ m which made up the sample. The PCB congeners tested for in this study were IUPAC numbers 28, 52, 101, 116, 153, 138, 170 and 180.

Using concentrations reported for bulk samples, concentrations for clav/silt samples, and percent size fractions of each sample, calculations were done to estimate concentrations on the greater than 63 µm portion of the sample. Similar calculations were done to determine organic carbon content on the >63 µm portion of the sample. Organic content varied from 0.01 percent to 10 percent inclusive of both <63 µm and >63 µm portions of the sediment.







Figure 18.—The organic carbon fractions (% dry weight) in five sediment size classes ranging from clay and sitt (<63 µm) to course sand (1.0 to 2.0 mm). Stations are indicated by hatch type. Data from [45].

19765

-





Figure 20 presents the percent organic carbon on the $<63 \,\mu\text{m}$ portion of the sample (filled bar) and on the $>63 \,\mu\text{m}$ portion of the sample (shaded bar). A comparison of the PCB congener concentrations on a dry weight basis (top) and on an organic carbon basis (bottom) is shown in Figure 21. Organic carbon basis (bottom) is shown in Figure 21. Organic carbon content in the $>63 \,\mu\text{m}$ class size at stations 2 and 4 is 0.01 percent and 0.06 percent respectively, as indicated in Figure 20. The data for these stations are shown on Figure 21 using filled symbols. Though an $f_{oc} > 0.2$ percent has been presented as the value for which organic carbon normalization applies, normalization at these foc values seems appropriate for this data set.

AND IN COMPANY

The top panel of Figure 21 indicates no evident relationship between PCBs in the <63 μ m sample and PCBs in the >63 μ m sample on a dry weight basis. When concentrations in either class size are normalized to organic carbon content then the concentrations are similar for both class sizes as shown in the bottom panel. This indicates that PCB concentrations are similar across sediment class sizes which supports organic carbon normalization.

Organic Carbon Fractiona







Organic Carbon Normalized



Figure 21.-Comparison of eight PCB congener concentrations of >63 µm sized particles, ordinate, to <63 µm sized particles, abscissa (Stations 1-7). Top panel is for dry weight normalization; bottom panel is for organic carbon normalization. PCB congeners are IUPAC Nos 28, 52, 101, 118, 138, 153, 179 and 180. Data from (47).

It can be concluded from the data of Prahl, Evans et al., and Delbeke et al., that the organic carbon-normalized PAH and PCB concentrations are relatively independent of particle size class and that organic carbon is the predominant controlling factor in determining the partition coefficient of the different sediment size particles in a sediment sample. The organic carbon concentration of the high-density sand-sized fraction in Prahl's data (0.2 to 0.3 percent) suggests that organic carbon normalization is appropriate at these low levels. The data from Evans et al. suggests that EqP can be applied to organic carbon originating from more than one source, that is, fragmentary plant matter and aging humic material.

Sediment/pore water partitioning. Normally when measurements of sediment chemical concentration, Cs, and total pore water chemical concentrations, Cpore, are made, the value of the apparent partition coefficient is calculated directly from the ratio of these quantities. As a consequence of DOC complexation, the apparent partition coefficient, Kp, defined as

$$K_{p}^{*} = \frac{C_{s}}{C_{pore}}$$
(23)

25

is given by

$$K_p^* = \frac{K_p}{1 + m_{\text{DOC}}K_{\text{DOC}}} = \frac{f_{\text{OC}}K_{\text{OC}}}{1 + m_{\text{DOC}}K_{\text{DOC}}}.$$
 (24)

As m_{DOC} increases, the quantity of DOC-complexed chemical increases and the apparent partition coefficient approaches

> (25) $K_{\rm p} = \frac{f_{\rm oc}K_{\rm oc}}{m_{\rm DOC}K_{\rm DOC}}$

which is just the ratio of sorbed to complexed chemical. Because the solid-phase chemical concentration is proportional to the free dissolved portion of the pore water concentration, Cd, the actual partition coefficient, Kp, should be calculated using the free dissolved concentration. The free dissolved concentration will typically be lower than the total dissolved pore water chemical concentration in the presence of significant levels of pore water DOC (e.g., Fig. 13). As a result, the actual partition coefficient calculated with the free dissolved concentration is higher than the apparent partition coefficient calculated with the total dissolved pore water concentration.

Direct observations of pore water partition coefficients are restricted to the apparent partition coefficient, Kp (Eqn. 23), because total concentrations in the pore water are reported and DOC complexing is expected to be significant at the DOC concentrations found in pore waters. Data reported by Brownawell and Farrington in 1986 [48] demonstrate the importance of DOC complexing in pore water. Figure 22 presents the apparent partition coefficient, measured for 10 PCB congeners at various depths in a sediment core, versus for Kow, the calculated partition coefficient. The line corresponds to the relationship, Koc = Kow, which is the expected result if DOC complexing were not significant. Because DOC concentrations were measured for these data, it is possible to estimate Ca with Equation 20 in the form:

$$C_{d} = \frac{C_{pere}}{1 + m_{DOC}K_{DOC}}$$
(26)



Figure 22.—Observed partition coefficient versus the product of organic carbon fraction and octanol/water partition coefficient. The line represents equality. The partition coefficients are computed by using total dissolved PCB (squares), and free PCB (circles) which is computed with Equation 26 with Kooc = Kow, Oata from (48).

and to compute the actual partition coefficient: $K_p = C_s/C_d$. The data indicate that if $K_{DOC} = K_{OW}$ is used, the results, shown on Figure 22, agree with the expected partition equation, namely that $K_p = f_{OC} K_{OW}$. A similar three-phase model was presented by Brownawell and Farrington in 1984 [49].

Other data with sediment/pore water partition coefficients for which the DOC concentrations have not been reported [50, 51] are available to assess the significance of DOC partitioning on the apparent sediment partition coefficient. Figure 23 presents these apparent organic carbon-normalized partition coefficients, that is $K_{0c} = K_{0}^{2} / f_{0c}$ versus K_{0w} . The expected relationship for DOC concentrations of 0, 1, 10, and 100 mg/L is also shown. Although there is substantial scatter in these data, reflecting the difficulty of measuring pore water concentrations, the data conform to DOC levels of 1.0 to 10 mg/L, which is well within the observed range for pore waters [40, 48]. Thus, these results do not refute the hypothesis that K_{0c} $\approx K_{0w}$ in sediments but show the need to account for DOC complexing in the analysis of pore water chemical concentrations.

Laboratory toxicity tests. Another way to verify Equation 22 is from data collected during sediment toxicity tests in the laboratory. These tests yield sediment ($C_{s,oc}$) and pore water (Ca) chemical toncentrations at several dosages bounding an experimentally estimated toxic concentration for the test organism. The organic content of the sediment must be measured also. Sediment toxicity tests are done under quiescent conditions and sediment and pore water are in equilibrium. The results of these tests can be used to





compute the organic carbon partition coefficient K_{oc} . To verify Equation 22, estimates of K_{oc} computed from Equation 11 using laboratory measurements of K_{ow} are then compared to partitioning in the sediment toxicity test. Sediment toxicity tests and K_{ow} measurements are available for five chemicals: endrin [20, 21, 52], dieldrin [54, 55], acenaphthene [56], phenanthrene [56], and fluoranthene [19, 57]. Sediment toxicity tests for these chemicals were performed as part of the development of SQC. Mortality results for these tests were presented in Figures 2 and 3.

Figure 24 shows organic carbon normalized sorption isotherm for acenaphthene, endrin, phenanthrene and fluoranthene, where the sediment concentration ($\mu g/g$ OC) is plotted versus pore water concentration ($\mu g/L$). These tests represent freshwater and marine sediments having a range of organic carbon content of 0.07 to 11.0 percent. In each

panel, the line corresponds to Equation 16 where K_{oc} is derived from K_{ow} measurements in the laboratory. A full discussion of laboratory K_{ow} measurements is presented subsequently. In each of the panels the toxicity test data are in agreement with the line computed from experimentally determined K_{oc} . For these cheming to DOC measurements are unavailable and partitioning to DOC in the pore water has not been considered. The figure indicates, however, that DOC complexing in these experiments appears to be negligible.

25

Partitioning in the dieldrin experiment indicated that DOC complexation may have been significant. The partitioning isotherm for dieldrin, Figure 25), represents organic carbon normalized sediment concentration, versus total (top panel) and computed dissolved (bottom panel) pore water concentrations. Dissolved pore water concentrations are computed



Figure 24.—Comparison of organic carbon partition coefficient (K_{oc}) observed in toxicity tests (symbols) to K_{oc} derived from laboratory K_{ow} and Equation 11 (solid line). Symbols are sediment concentration, ordinate, versus pore water concentration, abscissa. Solid line is $G_{s,oc} = K_{oc} \circ C_d$, where $Log_{10} \in K_{oc}$ is 3.76 for acenaphthene, 4.84 for endrin, 4.46 for phenanthrene, and 5.00 for fluoranthene. These Log₁₀ K_{oc} values are estimated from Log₁₀ K_{ow} values measured at the U.S. EPA Environmental Research Laboratory at Athens, Georgia. Data sources as indicated.



26

TOTAL PORE WATER CONCENTRATION (ug/L)



COMPUTED PORE WATER CONCENTRATION (ug/L)

Figure 25.—Comparison of organic carbon partition coefficient (H_{Oc}) observed in toxicity tests (symbols) to H_{Oc} derived from laboratory H_{Ow} and Equation 11 (solid line). Symbols are sediment concentration, ordinate, versus total (top panel) and free (bottom panel) pore water concentrations, abscissa. Solid line is $C_{s.oc} = H_{Oc} \circ C_d$, where Lógio H_{Oc} is 5.25 for dieldrin. The Logio H_{Oc} value is estimated from Logio H_{Ow} value measured at the U.S. EPA Environmental Research Laboratory at Athens, Georgia. Data source as indicated.

using Equation 26, DOC measurements and an estimated log $K_{DOC} = 5.25$. Log K_{DOC} is estimated from log $K_{oc} = 5.25$ for dieldrin. Figure 25 represents data from Hoke and Aukley [55] since Hoke [56] did not measure pore water. Adjusting for partitioning on to the DOC in the bottom panel results in better agreement with the experimentally determined Koc. These data represent one sediment with an organic carbon content of 1.6 percent. It is important to note that dieldrin has the highest K_{oc} of the five chemicals (Log₁₀ K_{oc} dieldrin = 5.25, acenaphthene = 3.76, endrin = 4.84, phenanthrene = 4.46, fluoranthene = 5.00. DOC complexing increases with an increasing partition coefficient which explains why DOC complexing is significant for dieldrin.

Organic Carbon Normalization of Biological Responses

The results discussed above suggest that if a concentration-response curve correlates to pore water concentration, it should correlate equally well to organic carbon-normalized total chemical concentration, independent of sediment properties. This is based on the partitioning formula $C_{s,oc} = K_{oc}Cd$ (Eqn.16), which relates the free dissolved concentration to the organic carbon-normalized particle concentration. This applies only to nonionic hydrophobic organic chemicals because the rationale is based on a partitioning theory for this class of chemicals.

Toxicity and bioaccumulation experiments. To demonstrate this relationship, concentration-response curves for the data presented in Figures 5 and 7 are used to compare results on a pore water-normalized and organic carbon-normalized chemical concentration basis. Figures 26 to 28 present these comparisons for kepone, DDT, endrin, and fluoranthene. The mean and 95 percent confidence limits of the LCso and ECso values for each set of data are listed in Table 2. The top panels repeat the responsepore water concentration plots shown previously in Figures 5 to 7, while the lower panels present the response versus the sediment concentration, which is organic carbon-normalized (microgram chemical per gram organic carbon).

The general impression of these data is that there is no reason to prefer pore water normalization over sediment organic carbon normalization. In some cases, pore water normalization is superior to organic carbon normalization, for example, keponemortality data (Fig. 26) whereas the converse sometimes occurs, for example kepone-growth rate (Fig. 26). A more quantitative comparison can be made with the LCsos and ECsos in Table 2. The variation of organic carbon-normalized LC305 and EC505 between sediments is less than a factor of two to three and is comparable to the variation in pore water LCsos and ECsos. A more comprehensive comparison has been presented in Figures 2 and 3, which also examine the use of the water-only LC50 to predict the pore water and sediment organic carbon LC505.

Bioaccumulation factors calculated on the basis of organic carbon-normalized chemical concentrations are listed in Table 3, for permethrin, cypermethrin, and





Figure 27.--Comparson of percent surwal of H. asteca to DDT (reft) and endin (right) concentration in pore water (top) and Organic Carbon Hormalized (ug/g os) (oo g'su) besitemtoli nodia0 oliepi0 00001 0001 001 01 10000 1000 001 Q L

3

n

30 ş

ÓP Ì

09

68

in bulk sediment, using organic carbon normalization (bottom) for three sediments with varying organic carbon concentra-

1008 (20' 3T)

Ō

30

07

00



28





kepone. Again, the variation of organic carbon normalized BAFs between sediments is less than a factor of two to three and is comparable to the variation in pore water BAFs.

Bioaccumulation and organic carbon normalization. Laboratory and field data also exist for which no pore water or DOC measurements are available but for which sediment concentration, organic carbon fraction, and organism body burden have been determined. These data can be used to test organic carbon normalization for sediments and to examine organism normalization as well. It is conventional to use organism lipid fraction for this normalization (see references in Chiou [58]). If Cb is the chemical concentration per unit wet weight of the organism, then the partitioning equation is

$$C_{b} = K_{i} f_{L} C_{d}$$
 (27)

where

- K_L = lipid/water partition coefficient (L/kg lipid)
- fL = weight fraction of lipid (kg lipid/kg organism)

 $C_d = free \text{ dissolved chemical concentration}$ ($\mu g/L$)

The lipid-normalized organism concentration, Cb.L. is

$$C_{b,L} = \frac{C_b}{f_L} = K_L C_d.$$
⁽²⁸⁾

The lipid-normalized body burden and the organic carbon-normalized sediment concentration can be used to compute a bioaccumulation ratio, which can be termed the BSF [59]:

$$BSF = \frac{C_{b,L}}{C_{s,oc}} = \frac{K_L}{K_{oc}} \approx \frac{K_L}{K_{ow.}}$$
(29)

The second equality results from using the partitioning Equations 16 and 28 and the third from the approximation that $K_{0r} = K_{0w}$. The BSF is the partition coefficient between organism lipid and sediment organic carbon. If the equilibrium assumptions are valid for both organisms and sediment particles, the BSF should be independent of both particle and organism properties. In addition, if lipid solubility of a chemical is proportional to its octancl solubility, $K_L \propto$ K_{0w} , then the lipid normalized-organic carbon normalized BSF should be a constant, independent of particles, organisms, and chemical properties [59, 60, 61]. This result can be tested directly.

The representation of benthic organisms as passive encapsulations of lipid that equilibrate with external chemical concentrations is clearly only a firstorder approximation. Biomagnification effects, which can occur via ingestion of contaminated food and the dynamics of internal organic carbon metabolism, can be included in a more comprehensive analysis [59]. It is, nevertheless, an appropriate initial assumption because deviations from the first-order representation will point to necessary refinements, and for many purposes this approximation may suffice.

A comprehensive experiment involving four benthic organisms—two species of deposit-feeding marine polychaetes, Nereis and Nephtys, and two species of deposit-feeding marine clams, Yoldia and Macoma—and five sediments has been performed by Rubinstein et al. [62]. The uptake of various PCB congeners was monitored until steady-state body burdens were reached. Sediment organic carbon and organism lipid content were measured. Figures 29 and 30 present the log mean of the replicates for the ratio of organism-to-sediment concentration for all measured congeners versus Kow for each organism. Dry weight normalization for both organism and sediment (left panels), organic carbon normalization for the sediment (center panels), and both organic





carbon and lipid normalization (right panels) are shown. The results for each sediment are connected by lines and separately identified.

The BSFs based on dry weight normalization are quite different for each of the sediments with the low carbon sediment exhibiting the largest values. Organic carbon normalization markedly reduces the variability in the BSFs from sediment to sediment (center panels). Lipid normalization usually further reduces the variability. Note that the BSFs are reasonably constant for the polychaetes, although some suppression is evident at log10Kow > 7. The clams, however, exhibit a marked declining relationship.

Results of a similar though less extensive experiment using one sediment and oligochaete worms have been reported [52]. A plot of the organic carbonand lipid-normalized BSF versus Kow from this experiment is shown on Figure 31, together with the averaged polychaete data (Fig. 29). There appears to be a systematic variation with respect to Kow, which suggests that the simple lipid equilibration model with a constant lipid-octanol solubility ratio is not descriptive for all chemicals. This suggests that a more detailed model of benthic organism uptake is required to describe chemical body burdens for all nonionic chemicals as a function of K_{ow} [59]. However, for a specific chemical and a specific organism, for example Nereis and any PCB congener (Fig. 29) organic carbon normalization reduces the effect of the varying sediments. This demonstrates the utility of organic carbon normalization and supports its use in generating SQC.

A further conclusion can be reached from these results. It has been pointed out by Bierman [63] that the lipid- and carbon-normalized BSF is in the range of 0.1 to 10 (Figs. 29 to 31) supports the contention that the partition coefficient for sediments is Koc = Kow and that the particle concentration effect does not appear to be affecting the free concentration in sediment pore water. The reason is that the lipid- and carbon-normalized BSF is the ratio of the solubilities of the chemical in lipid and in particle carbon (Eqn. 29). Because the solubility of nonionic organic chemicals in various nonpolar solvents is similar [64], it would be expected that the lipid-organic carbon solubility ratio should be on the order of one. If this ratio is taken to be approximately one, then the conclusion from the BSF data is that Koc is approximately equal to Kow for sediments [63].



Figure 30.—Plote of the 855 (ratio of organism-to-sediment concentration) for three sediments for a series of PCB congeners versus the logroffor for that congener. The dry weight normalization for both organism and sediment (left panels); organic carbon normalization for the sediment (middle panels); and drgr. and lipid normalization (right panels) as indicated. The orcarbon normalization for the sediment (middle panels); and drgr. and lipid normalization (right panels) as indicated. The orcarbon normalization for the sediment (middle panels); and drgr. and lipid normalization (right panels) as indicated if or gamerine are 'foldie (top) and Macome (pottom). Data from (G2), ¹/25



₽∠∠6τ

A final observation can be made. The data analyzed in this section demonstrate that organic carbon normalization accounts for much of the reported differences in bioavailability of chemicals in sediments for deposit-feeding polychaetes, oligochaetes, and clams. The data presented in previous sections are for amphipods and midges. Hence these data provide important additional support for organic carbon normalization as a determinant of bioavailability for different classes of organisms.

Determination of the Route of Exposure

The exposure route by which organic chemicals are accumulated has been examined in some detail for water column organisms (e.g., by Thomann and Connolly [65]). It might be supposed that the toxicity and bioaccumulation data presented above can be used to examine the question of the route of exposure. The initial observations were that biological effects appear to correlate to the interstitial water concentration, independent of sediment type. This has been interpreted to mean that exposure is primarily via pore water. However, the data correlate equally well with the organic carbon-normalized sediment concentration (see Figs. 2 and 3). This observation suggests that sediment organic carbon is the route of exposure. In fact, neither conclusion follows necessarily from these data because an alternate explanation is available that is independent of the exposure pathway.

Consider the hypothesis that the chemical potential or, as it is sometimes called, the fugacity [66] of a chemical controls its biological activity. The chemical potential, μ_d , of the free concentration of chemical in pore water, C_d, is

$$\mu_d = \mu_0 + RT \ln(Cd) \tag{30}$$

where μo is the standard state chemical potential and RT is the product of the universal gas constant and absolute temperature [67]. For a chemical dissolved in organic carbon—assuming that particle organic carbon can be characterized as a homogeneous phase—its chemical potential is

$$\mu_{oc} = \mu_{o} + RT \ln (C_{s,oc}) \tag{31}$$

where Cs,oc is the weight fraction of chemical in organic carbon. If the pore water is in equilibrium with the sediment organic carbon then

The chemical potential that the organism experiences from either route of exposure (pore water or sediment) is the same. Hence, so long as the sediment is in equilibrium with the pore water, the route of exposure is immaterial. Equilibrium experiments cannot distinguish between different routes of exposure.

The data analysis presented above, which normalizes biological response to either pore water or organic carbon-normalized sediment concentration, suggests that biological effects are proportional to chemical potential or fugacity.

The issue with respect to bioavailability is this: In which phase is μ most easily and reliably measured? Pore water concentration is one option.

However, it is necessary that the chemical complexed to DOC be a small fraction of the total measured concentration or that the free concentration be directly measured, perhaps by the C18 column technique [42]. Total sediment concentration normalized by sediment organic carbon fraction is a second option. This measurement is not affected by DOC complexing. The only requirement is that sediment organic carbon be the only sediment phase that contains significant amounts of the chemical. This appears to be a reasonable assumption for most aquatic sediments. Hence, SQC are based on organic carbon normalization because pore water normalization is complicated by DOC complexing for highly hydrophobic chemicals.

APPLICABILITY OF WQC AS THE EFFECTS LEVELS FOR BENTHIC ORGANISMS

The EqP method for deriving SQC utilizes partitioning theory to relate the sediment concentration to the equivalent free chemical concentration in pore water and in sediment organic carbon. The pore water concentration for SQC should be the effects concentration for benthic species.

This section examines the validity of using the EPA WQC concentrations to define the effects concentration for benthic organisms. This use of WQC assumes that (a) the sensitivities of benthic species and species tested to derive WQC, predominantly water column species, are similar, (b) the levels of protection afforded by WQC are appropriate for benthic organisms, and c) exposures are similar regardless of feeding type or habitat. This section examines the assumption of similarity of sensitivity in two ways. First, a comparative toxicological examination of the acute sensitivities of benthic and water column species, using data compiled from the published EPA WQC for nonionic organic chemicals as well as metals and ionic organic chemicals, is presented. Then a comparison of the FCVs and the chronic sensitivities of benthic saltwater species in a series of sediment colonization experiments is made.

19775

____31

Method-Relative Acute Sensitivity

32

The relative acute sensitivities of benthic and water column species are examined by using LC50s for freshwater and saltwater species from draft or published WQC documents that contain minimum database requirements for calculation of final acute values (Table 4). These data sets are selected because exposures were via water, durations were similar, and data and test conditions have been scrutinized by reviewing the original references. For each of the 2,887 tests conducted in fresh water, using 208 species with 40 chemicals, and the 1,046 tests conducted in salt water, using 118 species with 30 chemicals, the chemical, species, life stage, salinity, hardness, temperature, pH, acute value, and test condition (i.c., static, renewal, flow-through, nominal, or measured) were entered into a database. If necessary, original references were consulted to determine the tested life stage and any other missing information. Each life stage of the tested species was classified according to habitat (Table 5). Habitats were based on degree of association with sediment. A life stage that occupied more than one habitat was assigned to both of the appropriate habitats.

For each chemical, if a life stage was tested more than once or more than one life stage was tested, data were systematically sorted in a three-step process to arrive at the acute value based on the most experimentally sound testing methodology and the most sensitive life stages. First, if a life stage for a species was tested more than once, flow-through tests with measured concentrations had precedence, and data from other tests were omitted. When there were no flow-through tests with measured concentrations, all acute values for that life stage were given equal weight. If the remaining acute values for that life stage differed by greate: man a factor of four, the higher values were omitted and the geometric mean of the lower acute values was calculated to derive the acute value for that life stage. Second, life stages were classified as either "benthic" (infaunal species [habitats 1 and 2] or infaunal and epibenthic species [habitats 1, 2, 3, and 4]), or "water column" (habitats 5 to 8). Third, if two or more life stages were classified as either benthic or water column and their acute values differed by a factor of four, the higher values were omitted and the geometric mean of the lower acute values was calculated to derive the acute value for that life stage of the benthic or water column species. This procedure is similar to that used for WQC [8].

Comparison of the sensitivity of Benthic and Water Column Species

Most Sensitive Species. The relative acute sensitivities of the most sensitive benthic and water column species were examined by comparing the final acute values (FAV) for benthic and water column organisms, using acute LC50 concentrations from the 40 freshwater and the 30 saltwater WQC documents. When benthic species were defined as only Infaunal organisms (habitat types 1 and 2) and water column species were defined as all others (habitat types 3 to 8), the water column species were typically the most sensitive. The results are cross-plotted on Figure 32 (top). The line represents perfect agreement.

Data on the sensitivities of benthic infaunal species are limited. Of the 46 chemicals for which WQC for freshwater organisms are available, two or fewer infaunal species were tested with 28 (70 percent) of the chemicals, and five or fewer species were tested with 34 (85 percent) of the chemicals. Of the 30 chemicals for which WQC for saltwater organisms are available, 2 or fewer infaunal species were tested with 19 (63 percent) of the chemicals, and 5 or fewer species were tested with 23 (77 percent) of the chemicals. Of these chemicals only zinc in salt water has been tested using infaunal species from three or more phyla and eight or more families, the minimum acute toxicity database required for criteria derivation. As a result, FAVs could not be computed for several of the chemicals. Therefore, it is probably premature to conclude from the existing data that infaunal species are more tolerant than water column species.

A similar examination of the most sensitive benthic and water column species, where the definition of benthic includes both infaunal and epibenthic species (habitat types I to 4), is based on more data and suggests a similarity in sensitivity (Fig. 32, bottom). In this comparison, the number of acute values for freshwater benthic species for each chemical averaged nine, with a range of 2 to 27; the number of acute values for saltwater benthic species for each chemical substance averaged 11, with a range of 4 to 26. The variability of these data is high, suggesting that for some chemicals, benthic and water column species may differ in sensitivity and that additional testing is desirable, or that this approach to examining species sensitivity is not sufficiently rigorous.

Examination of individual criteria documents in which benthic species were markedly less sensitive than water column species suggests that the major factor for this difference is that benthic species phylogenetically related to sensitive water column species have not been tested. Apparent differences in sensitivity, therefore, may reflect an absence of appropriate data. Data that are available suggest that, on the average, benthic and water column species are similarly sensitive and support the use of WQC to derive SQC for the protection of infaunal and epibenthic species.

All species. A more general comparison of the species sensitivities can be made if all the LC30 data are used. One approach examines the relative loca-

,

. '33

.

	·		NO. OF 5/	LTWATER SPECIES	_	NO, OF F		-	
CHEMICAL	DATE OF PUBLICATION	TOTAL	INFAUNAL	EPIBENTHIC	COLUMN	TOTAL	INFANNAS		WATER
ACKNAPTHENE	9/87°	10		3	1			EPIBERINIC	COLUM
ACROLEIN	9/87°	-	-	-		12	1		<u> </u>
ADRIN	1980	16	, 0	11	12	21	<u>,+</u>	······································	<u> </u>
ALUMINUM	1986	-	-	_		15	<u>-</u>		15
AMMONIA	1985:1989	20	2	7	16	48	2		
ANTIMONY (III)	9/87°	11	3	6	5	9		<u>+</u> +	<u>ده</u> با
ARSENIC (III)	1985	12	2	3	8	16	<u>-</u> -	÷	
CADMIUM	1965	38	10	18	18	56		÷ •	13
CHLORDANE	1980	8	1	7	7	14		10	
CHLORINE	1985	23	2	9	15	33			10
OHLƏRPYRIFOS	1986	15	2		10	18			26
CHROMIUM (III)	1984				; <u>;-</u> -;			8	_11
CHROMIUM (VI)	1985	23	8			+			
COPPER	1985	25	6		1.10			10	21
CYANIDE	1785	9	1			17		+	36
PDT	1980	17	1	11		47	1	6	12
DIELDAIN	1980	21	1	15	15		3	15	29
2.4DIMETHYLPHENOL	6/88°	9	2	2		17	1		
ENDOSULFAN	1980	12	2			10			- 7
ENDRIN	1980	21	1	14	16	20 1	<u>-</u>		
IEPTACHLOR	1980	19	1	14	13	10 1		12	17
EXACHLOROCYCLOHEXANE	1980	19	2	14	17			8	12
EAD	1985	13	2						.10
AERCURY	1985	33	10		10		•	4	
ICKEL	1986	23	 !	10		30	11	8	12
ARATHION	1986				~~~~ +		2	7	13
ARATHION, METHYL-	10/88	- <u>-</u> -+			_ +	37		14	_23
ANTACHLOROPHENOL	1986	19	7		+	30	1	- 9	25
HENANTHRENE	9/87*	10 1	4			41		11	23
HENOL	5/889				 +		2	1	6
ELENIUM (IV)	1987	16	1 +		+		6		20
ELENIUM (VI)	1987					23	2	6	19
ILVER	9/879	21				12	1	4	10
HALLIUM	11/889	<u> </u>	· · · · · · · · · · · · · · · · · · ·			19	1	9	13
OXAPHENE	1986	15				8	1	ł	
RIBUTYLTIN	9/879					37	5	13	23
2.4-TRICHLOROBENZENF	9/86	16				9	1	1	6
4.8-TRICHLOROPHENDL	9/87	11 1				14	2		
IND	1007				5	10	1	2	8

* The total numbers of tested species may not be the same as the sum of the number of species from each habitat type because a species may occupy more than one habitat.

Dish aquatic life criteria document, U.S. Environmental Protection Agency, Office of Water Regulations and Standards, Criteria and Standards division, Washington, D.C.

			NO. OF SA	LTWATER SPECIES		HQ. OF FI	RESHWATER SPECIES	-	
CHEMICAL	DATE OF PUBLICATION	TOTAL	INFAUNAL	EPIBENTHIC	WATER COLUMN	TOTAL	INFAUNAL	EPIBENTHIC	WATER
ACEHAPTHENE	9/87*	10	-	3	1 7	1		<u>.</u>	2
ACROLEM	9/87°	-	-	-	; -	12	<u>t</u>	5	7
ADRIN	1980	16	0	11	12	21	2	10	15
ALUMINUM	1988	-	-	-	-	.15		5	11
AMMONIA	1985;1989	20	2	7	16	48	2	17	33
ANTIMOLOY (UII)	9/87 ^b	11	3	6	5	9		2	6
ARSYNIC (11)	1985	12	2	3	8	16	i	6	13
CADHIUM	1985	38	10	18	18	56	13	16	31
CHLORDANE	1980	8	1	7	7	14	1	4	10
CHLORINE	1985	23	2	9	15	33	1	9	26
CHLORPYRIFOS	1986	15	2	8	10	18	2	8	11
CHROMIUM (III)	1984	-	·			17	3	·	12
CHROMIUM (VI)	1 1985	23	8	. 9	9	33	· · · · · · · · · · · · · · · · · · ·	10	21
COPPER	1985	25	6	5	18	57		15	36
CYANIDE	1785	9	1	4	5	17	1	6	12
100	1980	17	1	11	12	42	3	15	29
DIELDRIN	1980	21	1	15	15	19		9	12
2.4DIMETHYLPHENOL	6/880	9	2	2	6	12	1	3	7
ENDOSULFAN	1980	12	2	8	8	10		4	7
ENDRIN	1980	21	1	14	16	28		12	17
HEPTACHLOR	1980	19	1	14	13	18	2	8	12
HEXACHLOROGYCLOHEXANE	1980	19	2	14	12	22	1	4	18
LEVD	1985	13	2	3	10	14		4	11
MERCUAY	1985	33	10		18	30	11	8	12
NICKEL	1986	23	7	10	9	$ _{21}^{-1} $	2	7	13
PARATHION	1986		-	- <u></u>	+	37	7	14	23
PARATHION, METHYL-	10/88		-		<u>† – – – – – – – – – – – – – – – – – – –</u>	36	1	9	25
PENTACHLOROPHENOL	1986	19	7	7	1 11	41	9	11	23
PHENANTHRENE	8/810	10	4	6	4	9	2	1	6
PHENOL	5/88°				+	32	6	1 9	20
BELENIUM (IV)	1987	16	1	5	1 13	23	2	 5	19
BELENIUM (VI)	1987	<u> </u>			<u> </u>	12	1	4	10
BILVER	9/870	21	<u> </u>	6	16	19	1		13
THAILIUM	11/889			· _	1				⊦
TOXAPHENE	1986	15	2	9	11	37		+	21
TAIBUTYLTIN	9/87ª	19		<u>-</u>	15	<u>⊢_</u> +	1	1	
1.2.4-TRICHLOROBENZENE	9/86	15	$\frac{1}{1}$		1	14		·	
2.4.5-TAICHLOROPHENOL	\$/87*	11		<u>-</u>	+	10	·•	÷;	
	1007	21	10	<u>-</u>	+		[*]		

The total numbers of tested species may not be the same as the sum of the number of species from each habital type because a species may occupy more than one habital.
 Draft equatic life criteria document, U.S. Environmental Protection Agency, Office of Water Regulations and Standards, Criteria and Standards division, Washington, D.C.

34

Sediment Quality Criteria Using Equilibrium Partitioning



Figure 32.—Comparison of LC₅₀ or EC₅₀ acute values for the most sensitive benthic (abscissa) and water column (ordinate) species for chemicals listed in Table 5. Benthic species are defined as infaunal species (habitat types 1 and 2, left panel) or infaunal and epibenthic species (habitat types 1–4); see Table 6.

tion of benthic species in the overall species sensitivity distribution. For each chemical in either fresh or salt water, one can examine the distribution of benthic species in a rank-ordering of all the species' LC505. If benthic species were relatively insensitive, then they would predominate in ranking among the larger LC50 concentrations. Equal sensitivity would be indicated by a u⁻¹ orm distribution of species within the overall ranking. Figure 33 presents the results for tests of nickel in salt water. The LC505 are plotted in rank order, and the benthic species are indicated. Infaunal species are among the most tolerant (left panel), whereas infaunal and epibenthic species (right panel). Table 5.—Habitat classification system for life stages of organisms.

HABITAT	DESCRIPTION
1.	Life stages that usually live in the sediment and whose food consists mostly of sediment or organisms living in the sediment; infaunal nonliter feeders.
2	Life stages that usually live in the sediment and whose food consists mostly of plankton and/or suspended organic matter filtered from the water column: infaunat filter feeders.
3	Life stages that usually live on the surface of sediment and whose food consists mostly of organic matter in sediments and/or organisms living in or on the sediment: epidenthic bottom feeders.
4	Life Stages that usually live on the surface of sediment and whose food is mostly from the water column, including suspended detritus, plankton, and larger prey: epidenthic water column feeders.
5	Life stages that usually live in the water column and whose food consists mostly of organisms that live on or in the sediment.
6	Life stages that usually live in and obtain their food from, the water column but have slight interaction with sediment because they occasionally rest or sit on the sediment and/or occasionally consume organisms that live in or on the sediment.
7	Life stages that live in or on such inorganic substrates as sand, rock, and gravel, but have negligible contact with sediment containing organic carbon.
8	Life stages that have negligible interactions with sediment because they spend essentially all their time in the water column and farely consume organisms in direct contact with the sediment; that is fouling organisms on plings, ships, and so on, and zooplankton, pelagic fish, and so on.

This comparison can be done chemical by chemical. However, to make the analysis more robust, the LCso data for each chemical-water type can be normalized to zero log mean and unit log variance as follows:

$$LC_{50n_i i j} = \frac{\log(LC_{50j j}) - \mu_i}{\sigma_i}$$
(33)

where i indexes the chemical-water type, μ_i is the log mean and si is the log standard deviation, j indexes the LC50s within the ith class, and LC50n,n is the normalized LC50. This places all the LC50s from each set of chemical-water type on the same footing. Thus, the data can now be combined and the uniformity of representation of benthic species can be examined in the combined data set.

The comparison is made in Figure 34. If the sensitivity of benthic species is not unique, then a constant percentage of benthic species-normalized LC505, indicated by the dashed line, should be repre-



Species Sensitivity for NI in Seawater







ien of Percentile Range (%)

Infaunal



Infaunal & Epibenthic



Figure 34.—Histograms of the proportion of saltwater and freshwater benthic organisms in 10 percentile groups of all normalized LCsos. If benthic organisms were as equally sensitive as water column organisms, the histograms should be of uniform height as indicated by the dashed line, the overall percentage of benthic species in the data set. Top panels include only infaunal organisms as benthic. The bottom panel includes infaunai and epibenthic as benthic organisms. 7

sented in each 10-percentile (decile) interval of data for all species. That is, the 10 rectangles in each histogram should be identical in height. The infaunal specie: top panel) display a tendency to be underrepresented in the lowest deciles. However, the infaunal and epibenthic species (boltom panels) more closely follow this idealized distribution. Infaunal and epibenthic freshwater species are nearly uniformly distributed, whereas the saltwater benthic species are somewhat underrepresented in the lowest ranks.

Given the limitations of these data, they appear to indicate that, except for possibly freshwater infaunal species, benthic species are not uniquely sensitive or insensitive and that SQC derived by using the FCV should protect benthic species.

Benthic Community Colonization Experiments

Toxicity tests that determine the effects of chemicals on the colonization of communities of benthic saltwater species (68-74) appear to be particularly sensitive at measuring the impacts of chemicals on benthic organisms. This is probably because the experiment exposes the most sensitive life stages of a wide variety of benthic saltwater species, and they are exposed for a sufficient duration to maximize response. The test typically includes three concentrations of a chemical and a control, each with 6 to 10 replicates. The test chemical is added to inflowing ambient seawater containing planktonic larvae and other life stages of marine organisms that can settle on clean sand in each replicate aquarium. The test typically lasts from two to four months, and the number of species and individuals in aquaria receiving the chemical are enumerated and compared to controls.

If this test is extremely sensitive and if concentrations in interstitial water, overlying water, and the sediment particles reach equilibrium, then the effect and no-effect concentrations from this test can be compared with the FCV from the saltwater WQC documents to examine the applicability of WQC to protect benthic organisms. An FCV is the concentration, derived from acute and chronic toxicity data, that is predicted to protect organisms from chronic effects of a chemical [8]. In addition, similarities in sensitivities of taxa tested as individual species and in the colonization experiment can indicate whether the conclusion of similarity of sensitivities of benthic and water column species is reasonable.

The benthic colonization experiment is consistent with the assumptions used to derive SQC. The initially clean sandy sediment will rapidly equilibrate with the inflowing overlying water chemical concentration as the pore water concentrations reach the overlying water concentration. The production of sedimentary organic matter should be slow enough to permit its equilibration as well. As a consequence, the organisms will be exposed to an equilibrium system with a unique chemical potential. Thus, the assumption of the EqP is met by this design. In addition, the experimental design guarantees that the interstitial water-sediment-overlying water is at the chemical potential of the overlying water. Hence there is a direct correspondence between the exposure in the colonization experiment and the wateronly exposures from which WQC are derived, namely, the overlying water chemical concentration. This allows a direct comparison.

Water Quality Criteria (WQC) Concentrations Versus Colonization Experiments

Comparison of the concentrations of six chemicals that had the lowest-observable-effect concentration (LOEC) and the no-observable-effect concentration (NOEC) on benthic colonization with the FCVs either published in saltwater portions of WQC documents or estimated from available toxicity data (Table 6) suggests that the level of protection afforded by WQC to benthic organisms is appropriate. The FCV should be lower than the LOEC and larger than the NOEC.

The FCV from the WQC document for pentachlorophenol of 7.9 μ g/L is less than the LOEC for colonization of 16.0 µg/L. The NOEC of 7.0 µg/L is less than the FCV. Although no FCV is available for Aroclor 1254, the lowest concentration causing no effects on the sheepshead minnow (Cyprinodon variegatus) and pink shrimp (Penaeus duorarum) as cited in the WQC document is about 0.1 µg/L. This concentration is less than the LOEC of 0.6 μ g/L and is similar to the NOEC of 0.1 μ g/L based on a nominal concentration in a colonization experiment. The lowest concentration tested with chlorpyrifos (0.1 μ g/L) and fenvalerate (0.01 μ g/L) affected colonization of benthic species. Both values are greater than either the FCV estimated for chlorpyrifos (0.005 µg/L) or the FCV estimated from acute and chronic effects data for fenvalerate (0.002 µg/L). The draft water quality criteria document for 1,2,4-trichlorobenzene suggests that the FCV should be 50.0 ug/L. This value is slightly above the LOEC from a colonization experiment (40.0 µg/L) suggesting that the criterion might be somewhat underprotective for benthic species. Finally, a colonization experiment with toxaphene provides the only evidence from these tests that the FCV might be overprotective for benthic species; the FCV is 0.2 µg/L versus the NOEC for colonization of 0.8 μ g/L.

The taxa most sensitive to chemicals, as indicated by their LC50s and the results of colonization experiments, are generally similar, although, as

19782

SUBSTANCE	COLONIZATION VERSUS FOVO	COHC. HELL	SENSITIVE TAXA	REFERENCE
	Colonization LOEC	16.0	Moliusas, Abundence	[69,70]
	FCV	7.9	Molluscs, Crustacea, Fish	
PENTACHLOROPHENOL	Colonization NOEC	7.0	-	
	Colorization LOEC	0.6	Crustacea	[71]
	Estimated FCV	-0.1	Crustacea, Fish	(68)
AROCLOR 1254	Colonization NOEC	0.1]-	1
	Colonization LOEC	0.1	Crustaces, Molluscs, species richness	[72]
	FCV	0.005	Crustacea	!
CHLOAPTRIFOS	Colonization NOEC		-	1
	Colonization LOEC	0.01	Crustacea, Chordates	[73]
	Estimated FCV	-0.002	Crustocea	
FENVALERAJE	Colonization NOEC		-	
	Estimated FCV	50.	Crustøcea, Fish	[74]
	Colonization LOEC	40.	Molluscs, abundance	!
1,2,4-TRICHLOROBENZENES	Colonization NOEC		-	
	Colonization LOEC	11.0	Crustacea, species richness	(68)
	Colonization NOEC	0.8	-	ļ — —
TOXAPHENE	FCV	0.2	Crustacea, fish	1

Table 6.—Comparison of WQC FCVs and concentrations affecting (LOEC) and not affecting (NOEC) benthic colonization.

*Six day exposure to established benthic community

might be expected, differences occur. Both the WQC documents and the colonization experiments suggest that crustacea are most sensitive to Aroclor 1254, chlorpyrifos, fenvalerate, and toxaphene. Colonization experiments indicated that molluscs are particularly sensitive to three chemicals, an observation noted only for pentachlorophenol in WQC documents. Fish, which are not tested in colonization experiments, are particularly sensitive to four of the six chemicals.

Conclusions

Comparative toxicological data on the acute and chronic sensitivities of freshwater and saltwater benthic species in the ambient WQC documents are limited. Acute values are available for only 34 freshwater infaunal species from four phyla and only 28 saltwater infaunal species from five phyla. Only seven freshwater infaunal species and 24 freshwater epibenthic species have been tested with five or more of the 40 WQC chemicals. Similarly, nine saltwater infaunal species and 20 epibenthic species have been tested with five or more of the 30 substances for which saltwater criteria are available.

In spite of the paucity of acute toxicity data on benthic species, available data suggest that benthic species are not uniquely sensitive and that SQC can be derived from WQC. The data suggest that the most sensitive infaunal species are typically less sensitive than the most sensitive water column (epibenthic and water column) species. When both infaunal and epibenthic species are classed as benthic, the sensitivities of benthic and water column species are similar, on average. Frequency distributions of the sensitivities of all species to all chemicals indicate that infaunal species may be relatively insensitive but that infaunal and epibenthic species appear almost evenly distributed among both sensitive and insensitive species overall.

Finally, in experiments to determine the effects of chemicals on colonization of benthic saltwater organisms, concentrations affecting colonization were generally greater, and concentrations not affecting colonization were generally lower, than estimated or actual saltwater WQC FCVs.

GENERATION OF SQC

Parameter Values

The equation from which SQC are calculated is

$$SQC_{oc} = K_{oc}FCV$$
 (34)

(see Eqns. 2 to 7 and associated text). Hence, the SQC concentration depends only on these two parameters. The K_{oc} of the chemical is calculated from the K_{ow} of the chemical via the regression Equation 11.

-38

Sediment Quality Criteria

The reliability of SQCoc uppends directly on the reliability c. Kaw. For most chemicals of interest, the available Kows (e.g., [75]) are highly variable—a range o wo orders of magnitude is not unusual. Therefore the measurement methods and/or estimation methodologies used to obtain each estimate must be critically evaluated to ensure their validity. The technology for measuring Kow has improved in recent years. For example, the generator column [76] and the slow stirring [73] method appear to give comparable results, whereas earlier methods produced more variable results. Hence, it is recommended that literature values for Kows not be used unless they have been mounted using the newer techniques.

Measurement of Kow

The Kow is defined as the ratio of the equilibrium concentrations of a dissolved substance in a system consisting of n-octanol and water and is ideally dependent only on temperature and pressure:

 $K_{\rm ow} = C_{\rm OCT}/C_{\rm W}$ (35)

where C_{OCT} is the concentration of the substance in n-octanol and C_w is the concentration of the substance in water. The K_{ow} is frequently reported in the form of its logarithm to base ten as log P.

At the EPA Environmental Research Laboratory (ERL) at Athens, Georgia, three methods were selected for measurement and one for estimation of K_{ow} for the five chemicals for which SQC are being developed. The measurement methods were shakecentrifugation (SC), generator column (GC), and slow-stir-flask (SSF). The estimations were made using the computer expert system SPARC. The discussion of these methods is adapted from ERL at Athens research protocols.

The SC method [78] is routinely used to measure the partitioning of compounds with K_{ow} values on the order of 10° to 10° . The method involves adding a layer of octanol containing the compound of interest onto the surface of the water contained in a centrifuge tube. Both phases were mutually presaturated before beginning the measurements. Equilibration is established by gentle agitation and any emulsions formed are broken by centrifugation. The concentration in each phase is determined, usually b'' a chromatographic method, and the K_{ow} value calculated using Equation 35.

The original GC method, limited to compounds with K_{ow} values of less than 10⁶, was modified [76] and used to determine K_{ow} values up to 10⁸. Briefly, the method requires the packing of a 24-cm length of tubing with silanized Chromosorb W. Octanol, containing the chemical in a known concentra-

ning

Equilibrium Parti

tion, is then pulled through the dry support by gentle :ction until the octanol appears at the exit of the col-...m. Water is then pumped through the column at a rate of less than 2 mL per minute to allow equilibration of the chemical between the octanol and water. The first 100 mL are discarded followed by collection of an amount of water sufficient to determine the chemical concentration. The Kow is calculated using Equation 35.

The SSF method [77] achieves equilibrium of the compound between octanol and water by a gentle stirring of the phases contained in a six-liter flask. One liter aliquots of the aqueous phase are withdrawn at two-day intervals and the concentration of the chemical determined. Equilibrium is considered to e established when the concentration of the chemical is constant in successive samples (usually after two to six days). The procedure is to set up three six-liter flasks in a constant temperature room. Five liter of water are added to each flask and the water is surred with teflon-coated magnetic stir bars overnight to achieve temperature equilibration. The temperature equilibrated octanol is added very gently along the side wall to avoid mixing of the two phases. At the time of sampling, a one-liter aqueous sample is drained from a sampling port at the base of the flask without disturbing the octanol layer. The concentration in each phase is determined, usually by a chromatographic method, and the Kow value calculated using Equation 35.

When repetitive measures were made in the Athens laboratory, a protocol was established to assure compatibility with future experiments. These protocols described the entire experimental scheme including planning, sample requirements, experimental set up and chemical analysis, handling of data, and quality assurance. Only established anaiytical methods for solute concentration measurement were applied and the purity and identity of the chemical was determined by spectroscopic means. The name on the label of the chemical's container was not proof of identity.

Standard reference compounds (SRCs) were tested with each experiment. SRCs are compounds that are used as quality assurance standards and as references in inter-laboratory generation of data. The value of the process constant(s) has been established by repetitive measurements for an SRC and serves as baseline information for evaluating all experimental techniques and all aspects of quality assurance. Because the SRC is taken through the entire experimental scheme, its acceptable result assures the experimenter that equipment and measurement methods are functioning satisfactorily. Table 7 shows the log10Kow values for endrin, dieldrin, acenapthene, phenantherene, and fluoranthene and the

IEMICAL	\$C	QC	85 F
ENDRIN	4.65	4.67	4.86
	4.91	5.01	4,59
L	4.79	4.73	4.97
	4.76	4.62	4.95
	4.84	5.09	5.02
	4.83	5.28	4.82
	4.84	·	5.04
	4.83		4.91
			5.07
			4.93
		•	4.98
		•	4.79
DIELORIN	5.04	4.89	5.33
	5.00	4.88	5.43
	5.04	5.18	5.38
	5.03	\$.15	5.33
	5,04	5.26	5.43
	4.86	5.38	5.08
	4.99	5.67	5.28
	5.04	·	
AGENAPHTHENE	3.82	4.18	3.81
	3.84	4.17	3.84
	3.68	4.16	3.84
	3.84	4.17	
PHENANTHRENEL	4.29	4.47	4.57
<u> </u>	4.25	4.41	4.53
	4.33	4.46	4.50
	4.33	4.24	
FLUORANTHENE	4.99	5.19	4.98
	5.00	5.35	5.02
	5.01	5.47	5 02
		5.48	5.10
[· · · · · · · · · · · · · · · · · · ·	5.14
		· · · · · · · · · · · · · · · · · · ·	5.23
BIPHENYL	4.06	1	
		······································	1

Table 7 .- Log10Kow values measured by shake-centrifugation (SC), Generator column (GC), and

Source: U.S. EPA Environmental Research Laboratory, Athens Georgia.

SRCs, biphenyl and pyrene, measured at the Athens laboratory by the SC methods. The SRCs were not measured by the GC or SSF methods.

The log₁₀ of the average of eight previous measurements of K_{ow} by the shake-centrifugation method for biphenyl is 4.06. The log₁₀ of the average of 13 previous measurements of K_{ow} by the shake-

centrifugation method for pyrene is 5.17. These average Kows are in good agreement with the SQC shakecentrifugation measurements for biphenyl and pyrene made concurrently with the measurements for the five chemicals providing quality assurance for the experimental techniques (see Table 7).

literature.

Literature Kow. An extensive literature search was performed for the five compounds and two standard reference compounds, biphenyl and pyene. Generally, problems encountered in compiling and reporting fate constants from published data and from databases during a several years have ranged from retrieval of misquoted numbers to solution of nested citations [79]. Some citations were three or more authors removed from the original work or contained data referenced as unpublished data or as personal communication. The same problems were experienced during a ERL, Athens literature search. The largest difference in misquoting numbers was six orders of magnitude. For these reasons, ERL obtained data from the primary sources and released values only from these primary sources. Unpublished data or data that originated through personal communication were rejected as well as data that were insufficiently documented to determine their credibility and applicability or reliability.

Tables 8 and 9 show the measured and estimated log10Kow values, respectively, retrieved by this literature search. Each of the measured values was experimentally determined by the researcher using one of several laboratory methods. The individual experimental methods are not identified here. The estimated literature values were computed by the researchers by one of several published techniques. The individual computational techniques also are not identified here.

Estimated Kow. A promising new computational method for predicting reactivity is the computer expert system SPARC (SPARC Performs Automated Reasoning in Chemistry) being developed by ERL's Samuel W. Karickhoff and scientists at the University of Georgia [106]. The system has the capability of crossing chemical boundaries to cover all organic chemicals and uses algorithms based on fundamental chemical structural theory to estimate parameters. Organic chemists have in the past established the types of structural groups or atomic arrays that impart certain types of reactivity and have described, in "mechanistic" terms, the effects on reactivity of the structural constituents appended to the site of reaction.

To encode this knowledge base, Karickhoff and his associates developed a classification scheme that defines the role of structural constituents in affecting or modifying reactivity. SPARC quantifies reactivity by classifying molecular structures and selecting appropriate "mechanistic" models--lt uses an approach that combines principles of quantitative structure-activity relationships, linear free energy theory (LFET), and perturbed molecular orbital (PMO) or quantum chemistry theory. In general, SPARC uses LFET to compute thermal properties and PMO theory to describe quantum effects such ar delocalization energies or polarizabilities of *pi* electrons.

CHEMICAL	LOGIOKOW VALUE	REFERENCE
ENDRIN	4.40	(80)
	4.92	[81]
	5.01	(82)
	5.195	[77]
DIELDRIN	4.09	[83]
1	4.54	(63)
	4.65	[84]
	5.401	[77]
	6.2	[85)
AGENAPTHENE	3.92	(86)
PHENANTHRENE	4.28	[87]
	4.46	(88)
	. 4.562	(77)
	4.57	[78]
·····	4.63	(89)
FLUORANTHENE	5.155	(77)
BIPHENYL	3.16	(90)
	3.63	(84)
	3.75	(91)
	3.76	[92]
	3.79	[80]
1	3,89	[76]
	4.00B	(77)
	4.01	(82)
	4.04	[66]
	4.09	[84]
	4.10	[89]
PYRENE	4.96	[80]
	5.05	{84}
	5.09	[93]
	5.18	(78)
!	5.22	(69)

Table 8 .- Measured log10 Kew values found in the

SPARC computes Kow values from activity coefficients in the octanol (~1o) and water (~1w) phases using Equation 36:

 $LogioKow = logio(-1w/-1_0) + logio(Mo/Mw) (36)$

5.52

1

(94)

where M₀ and M_w are solvent molecularities of octanol and water, respectively. SPARC computes activity coefficients for any solvent/solute pair for which the structure parser can process the structure codes. Ultimately, any solvent/solute combination can be addressed. New solvents can be added as easily as solutes by simply providing a Simplified Molecular

40

CHEMICAL	LOOLOKOW VALUE	REFERENCE
ENDAIN	3.54	(95)
	5.6	[96]
DIELDAIN	3.54	(95)
ACENAPTHENE	3.70	(96)
	3.92	[97]
	3.98	[95]
	4.03	(98)
	4.15	[99]
	4.22	(100)
j	4,33	(101)
	4.43	[102]
PHENANTHRENE	4.44	(100)
	4.45	(95)
	4.63	(99)
	4.84	(96)
FLUORANTHENE	4.90	[95]
	4.95	U.S. EPA, Graph Exposure Model System (GEMS)
	5.22	[96]
	5.29	(99)
	5.33	[101]
BIPHENYL	3.79	[96]
	3.95	(97)
	3.98	(100)
	4,14	(99)
	4.25	(107)
	4.42	(103)
PYRENE	4.50	[104]
	4.85	(100)
	4.88	[105]
	4.90	(95)
	5.12	[99]
	5.22	(96)
	5.32	(101)

* The Graphical Exposure Modeling System (GEMS) is an interactive computer system located on the VAX Cluster in the National Computer Center in Research Triangle Park, North Cerolina, under management of EPA's Office of Toxic Substances. PC GEMS is the version for personal computers.

Interactive Linear Entry System (SMILES) string [107, 108]. Activity coefficients for either solvent or solute are computed by solvation models that are built from structural constituents requiring no data besides the structures.

A goal for SPARC is to compute a value that is as accurate as a value obtained experimentally for a fraction of the cost required to measure it. Because SPARC does not depend on laboratory operations conducted on compounds with structures closely related to that of the solute of interest, it does not have the inherent problems of phase separation encountered in measuring highly hydrophobic compounds (log10Kow > 5). For these compounds, SPARC's computed value should, therefore, be more reliable than a measured one. Reliable experimental data with good documentation are still necessary, however, for further testing and validation of SPARC.

CLOGP [109] is a computerized program that estimates the log10Kow, based on Leo's Fragment Constant Method [105]. CLOGP provides an estimate of log10Kow using fragment constants (f_i) and structural factors (F_i) that have been empirically derived for many molecular groups. The estimated log10Kow is obtained from the sum of constants and factors for each of the molecular subgroups comprising the molecule using Equation 37.

> $Log_{10}K_{ow} = \Sigma (f_i + F_i)$ i = 1

(37)

41

The method assumes that logicKow is a linear additive function of the structure of the solute and its constituent parts and that the most important structural effects are described by available factors. The structure of the compound is specified using the SMILES notation. The CLOGP algorithm is included in the database QSAR (see Table 10) located at EPA's Environmental Research Laboratory at Duluth, Minnesota. All CLOGP values reported here were obtained through QSAR. Table 10 shows the estimated logitoKow values that were computed with SPARC and CLOGP.

Table 10.—QSAR *obtained Log10Kow values estimated by SPARC and CLOGP

CHEMICAL	SPARC	CLOGA
ENDRIN	5.40	
DIELDRIN	5.40	-
ACENAPTHENE	3.88	4.07
PHENANTHRENE	4.58	4,49
FLUORANTHENE	5.21	4.95
BIPHENYL	4.25	4.03
PYRENE	5.13	4.95

•Quantitative Structure Activity Relationships (QSAR) is an interactive chamical database and hazard assessment system designed to provide basic information for the evaluation of the faile and effects of chemicals in the environment. QSAR was developed jointly by the U.S. EPA Environmental Research Laboratory, Duluth, Minnepota, Montana State University Center for Data System and Analysis, and the Pomone College Medicinal Chemistry Project.

 K_{ow} Selection. Investigators selected SSF derived Kow values to derive the Koc to calculate SQC concentrations because SSF is the superior method for chemicals with low and high Kow values, has the least statistical bias, and is highly reproducible. Use of values from one method provides consistency across chemicals. This choice was made after an analysis of the Kow values generated by the three measurement methods discussed above (GC, SC, and SSF) and the SPARC estimation method for the five chemicals for which SQC are currently being developed (acenaphthene, dieldrin, endrin, fluoranthene, and phenanthrene). Kows were measured with repeat experiments and a mean Kow was computed for each method, for each chemical (Table 11).

42

The mean measured Kows and the SPARC estimatio. nethod provide similar Kow estimates. To select a mal Kow for computing the SQC, the four methods were compared and the bias of each method were quantified. (Bias is defined as the mean difference between the best-fit estimate of Kow using all four methods and the estimates from each method.) Figure 35 presents the mean measured Kows for each chemical and the range of values. SC tends to estimate lower values while the GC method estimates higher values. GC values exhibit greater variability than the SC and SSF values. SSF estimates of Kows were generally within the range of the SC and GC methods.

A statistical analysis of the three measurement methods and SPARC method was performed. The following linear model was used to compute estimates of K_{0W} for each chemical (represented by E1, E2, E3, E4, E5) and the biases contributed by each of the estimation methods (represented by B1, B2, B3, B4). The regression model is as follows:

Og10Kow	8	E1 •	Endrin +
		E2 •	Dieldrin +
		E3 •	Acenaphthene +
		E5 •	Fluoranthene +

B1 * Shake Centrifugation +

82 * Generator Column +

B3 • Slow-Stir-Flask B4 • SPARC To compute \log_10K_{ow} the variables, ENDRIN, ... SPARC, are set to either 0 or 1 and the appropriate coefficient for the chemical and method that corresponds to the particular K_{ow} measurement or estimate is selected. Table 12 presents the model K_{ow} results and bias contributed by each method. Shake centrifugation and SPARC estimates provide the greatest bias, followed by the generator column method. Slow-stir-flask provides the least bias. Slowstir-flask was chosen as the method to use to determine K_{ow} for use in computing SQC since it appears to have the least bias. In addition, it exhibits similar variability to the shake centrifugation method and less variability than the generator column method.

Table 12.--Model results to determine method bias

COEFFICIENT	ESTIMATE OF Kow OR BIAS FOR	VALUE
E1	Endrin	4.88
£2	Dielgnn	5.17
E3	Acenaphthene	3.94
E4	Phenanthrane	4.40
ξ5	Fluoranthene	5.13
91	Shake Centrilugation	-0.115
82	Generator Column	0.091
83	Slow-sur flask	0.040
84	SPARC	0.193

 K_{oc} Determination. The previous section discusses selecting the method for measuring Kows for use in computing SQC. It is widely accepted that K_{ocs} can be estimated from K_{ow} . The K_{oc} used to calculate the sediment quality criteria is based on the regression of log10Koc to log10Kow, Equation 11.

Logio Kow (number of determinations)					
CHEMICAL	SLOW STIR FLASH				
ENDRIN	I 4.80 (8)	4.97 (6)	4.92 (12)		
DIELDRIN	5.01 (8)	5.16 (7)	5.34 (7)		
Acenaphthene	3.85 (4)	4.17 (4)	3.83 (3)		
PHENANTHRENE	4.30 (4)	4.40 (4)	4.54 (3)		
FLUDRANTHENE	5.00 (3)	5.39 (4)	5.09 (6)		

Sediment Quality Criteria Using Equilibrium Partitioning 9 9 9 1 ACENAPHITHENE ENDRIN PHEMANTHRENE DIELDRIN FLUCRANTHREE

Figure 35.—Laboratory K_{ow} values for five chemicals using three experimental methods with replication. K_{ow} values were measured at the Environmental Research Laboratory. Athens, GA, for each chemical the average of the methods is indicated by 0 for shake centrifugation, \Box for slow stir flask and \sim for generator column. Ranges are indicated by Σ .

This equation is based on any analysis of an extensive body of experimental data for a wide range of compound types and experimental conditions, thus encompassing a wide range of K_{ows} and f_{ocs} .

Sediment toxicity tests provide a favorable environment for measuring Kow. Figures 24 and 25 presented plots of the organic carbon-normalized sorption isotherm from sediment toxicity tests for the five chemicals where the sediment concentration $(\mu g/g \circ c)$ is plotted versus pore water concentration ($\mu g/L$), Also included in each panel is the line to the partition, Equation 16, where Koc is computed from the slow-stir flask Kow values. These plots can be used to compare the Koc computed from laboratory determined Kow and the regression equation with the partitioning behavior of the chemical in the sediment toxicity tests. For each of the chemicals the Koc line is in agreement with the data demonstrating the validity of the use of the slow-stir flask Kow in the SQC computation.

Species Sensitivity

The FCV is used as the appropriate end point for the protection of benthic organisms. Therefore, its applicability to benthic species for each chemical should be verified. The previous work has indicated that this is a reasonable assumption across all criteria chemiccals. To test this assumption for a particular chemical a statistical method known as Approximate Randomization [110] is applied to each chemical. The idea is to test whether the difference between the final acute value (FAV) derived from considering only benthic organisms is statistically different from the FAV contained in the Water Quality Criteria (WQC).

The Approximate Randomization method tests the significance level of the test statistic by comparing it to the distribution of statistics generated from many random reorderings of the LCso values. For example, the test statistic in this case is the difference between the WQC FAV, computed from the WQC LC50 values, and the benthic FAV, computed from the benthic organism LC50 values. Note that the benthic organism LC50 values are a subset of the WQC LCso values. In the Approximate Randomization method for this test, the number of data points coinciding with the number of benthic organisms are selected from the WQC data set. A "benthic" FAV is computed. The original WQC FAV and the "benthic" FAV are then used to compute the difference statistics. This is done many times and the distribution that results is representative of the population of FAV difference statistics. The test statistic is compared to this distribution to determine its level of significance.

For each chemical, an initial test of the difference between the freshwater and saltwater FAVs for all species (water column and benthic) is performed. The probability distribution of the FAV differences for fluoranthene are shown in the top panel of Figure 37. The horizontal line that crosses the distribution is the test statistic computed from the original WQC and benthic FAVS. For fluoranthene, the test statistic falls at the 78th percentile. Since the probability is less than 95 percent, the hypothesis of no significant difference in sensitivity is accepted.

Since freshwater and saltwater species show similar sensitivity, a test of difference in sensitivity for benthic and WQC organisms combining freshwater and saltwater species can be made. The bottom panel of Figure 36 represents the bootstrap analysis to test the hypothesis of no difference in sensitivity between benthic and WQC organisms for fluoranthene. The test statistic for this analysis falls at the 74th percentile and the hypothesis of no difference in sensitivity is accepted.

10

Table 13 presents the Approximate Randomization analysis for five chemicals for which SQC documents are being developed. Four chemicals, (acenaphthene, phenanthrene, fluoranthene and dieldrin) indicate that there is no difference in sensitivity for freshwater and saltwater species. The test for endrin fails at the 99 percentile which indicates that FAVs for freshwater and saltwater are different. Therefore, separate analyses for the freshwater and saltwater organisms are performed.

Table 14 presents the results of the statistical analysis for each chemical for benthic organisms and WQC organisms. In all cases the hypothesis of no difference in sensitivity is accepted. Therefore, for each individual chemical the WQC is accepted as the appropriate effects concentrations for benthic organisms.

Quantification of Uncertainty Associated with SQC

The uncertainty in the SQC can be estimated from the degree to which the equilibrium partitioning model, which is the basis for the criteria, can rationalize the available sediment toxicity data. The EqP model asserts that (1) the bioavailability of nonionic organic chemicals from sediments is equal on an organic carbon basis; and (2) that the effects concentration in sediment can be estimated from the product of the effects concentration from water-only exposures and the partition coefficient K_{0c} . The uncertainty associated with the sediment quality criteria can be obtained from a quantitative estimate of the degree to which the available data support these assertions.



i.

:;



The data used in the uncertainty analysis are the water-only and sediment toxicity tests that have been conducted in support of the sediment criteria development effort. A listing of the data sources used in the EqP uncertainty analysis is presented in Table 15. These freshwater and saltwater tests span a range of themicals and organisms; they include both wateronly and sediment exposures, and they are replicated within each chemical-organism-exposure media

	NU	NUMBER		FINAL ACUTE VALUE (FAV)		
OHEMICAL	SALT WATER	FRESH WATER	SALT WATER	FRESH WATER	DIFF	*
EMORIN	19	35	0.033	0.189	0.156	99
DIELDRIN	21	19	0.662	0.377	-0.305	31
AGENAPHTHENE	10	10	139.05	80.01	-59.04	33
PHENANTHRENE	-11	8	16.8	59.63	43.02	73
FLUORANTHENE	8	12	16.13	33.58	17.45	78

		NUMBER		FINAL ACUTE VALUE (FAV)			
CHEMICAL	WATER TYPE	WATER COL + BENTHIC	BENTHIC	WATER COL + BENTHIC	BENTHIC	- D1FF	
ENDRIN	fieln	35	21	0.189	.234	-0.045	7
ENDRIN	Salt	19	12	0.033	0 023	0.010	66
DIELDAIN	Combined	40	26	0.621	0.532	0.090	72
CENAPHTHENE	Combined	20	10	132.6	173.9	-38.34	31
HEHANTHRENE	Combined	19	13	26.62	19.27	7.35	80
LUORANTHENE	Combined	20	14	37.91	34.27	3.64	74

ENDRIN	DIELD	FLUGR	ACENAP	ACENAP	PHEN	PHEN
Hy	Ну	Re	Le	Εο	Le	Εo
3(3/-)	1/-	1/-	4/-	4/.	4/-	4/-
2/11%	2/1.7%	2/.18%	2/1.6%	2/1.2%	2/1.9%	2/1.0%
4/11%	4/2.9%	2/.30%	2/2.5%	2/2.5%	2/2.5%	2/2.5%
3/3.0%	4/8.7%	2/.48%	2/3 6%	2/3.6%	2/3.6%	2/3.3%
3/6.1%				i — — — — — — — — — — — — — — — — — — —		,
9/11.N						
/ # Hyalella	Re = Rhepoxynlue	Le = Leptocheirve	Eo a Echaus	torius		

Data sources: endin (20, 21, 52), dicidin (53) fluoranthene (19), acchaphtnene (56) phenanthrene (56).

treatment. These data are analyzed using an analysis of variance (ANOVA) to estimate the uncertainty (i.e., the variance) associated with varying the exposure media and the uncertainty associated with experimental error. If the EqP model were perfect, then there would be only experimental error. Therefore, the uncertainty associated with the use of EqP is the variance associated with varying exposure media.

Sediment and water only LCsos are computed from the sediment and water-only toxicity tests. The EqP model can be used to normalize the data in order to put it on a common basis. The LCso from wateronly exposures, LC50w (Hg/L), is related to the LC50 for sediment on an organic carbon basis, LC505,oc (ng/goc) via the partitioning equation:

LC50s,oc = Koc LC50w (38)

The EqP model asserts that the toxicity of sediments expressed on an organic carbon basis equals toxicity in water-only tests multiplied by the Kee. Therefore, either LCsos, oc ($\mu g/goc$), from sediment toxicity experiments or Koc x LCsow ($\mu g/goc$), are estimates of the true LCso for this chemical-organism pair.

In this analysis, the accuracy of Koc is not treated separately. Any error associated with Koc will be reflected in the uncertainty attributed to varying the exposure media.

In order to perform an analysis of variance, a model of the random variations is required. As discussed above, experiments that seek to validate Equation 38 are subject to various sources of random variations. A number of chemicals and organisms have been tested. Each chemical-organism pair was tested in water-only exposures and in different sediments. Let a represent the random variation due to the varying exposure media. Also, each experiment was replicated. Let c represent the random variation due to replication. If the model were perfect, there would be no random variation other than that resulting from the experimental error which is reflected in the replications. Thus, a represents the uncertainty due to the approximations inherent in the model, and ϵ represents the experimental error. Let σ_0^2 and σ_{ϵ}^2 be the variances of these random variables. Let i index a specific chemical-organism pair. Let j index the

19791

exposure media, water-only, or the individual sediments. Let k index the replication of the experiment. Then the equation that describes this relationship is

46

$$Ln(LC_{50i,j,k}) = \mu_i + \alpha_{i,i} + \varepsilon_{i,j,k}$$
(39)

where $Ln(LC_{50|1,1,k})$, are either $ln(LC_{50w})$ or $ln(LC_{50pw})$ corresponding to a water-only or sediment exposure; μ_{13} are the population of $ln(LC_{50})$ for chemical-organism pair i. The error structure is assumed to be lognormal which corresponds to assuming that the errors are proportional to the means, for ex. $\neg le$, 20 percent; rather than absolute quantities, for example, 1 mg/L. The statistical problem is to estimate μ_1 and the variances of the model error, σ_a^2 , and the measurement error, σ_t^2 . The maximum likelihood method is used to make these estimates [109]. The results are shown in Table 16.

Table 16.—Analysis of variance for derivation of criteria confidence limits.			
SOURCE OF UNCERTAINTY	VALUE (HE/goc)		
Exposure medie	1 0 ₀	0.39	
Replication	0,	0.21	
Sediment Quality Criteria	0500	0.39	

Note that $\sigma_{SQC}=\sigma_{\alpha}$ the variability due to EqP

The last line of Table 16 is the uncertainty associated with the SQC, that is, the variance associated with the exposure media variability. The confidence limits for the SQC are computed using this uncertainty for SQC. For the 95 percent confidence interval limits, the significance level is 1.96 for normally distributed errors. Thus

 $\ln(SQC_{oc})_{upper} = \ln(SQC_{oc}) + 1.96\sigma_{SQC}$ (40)

 $In(SQC_{oc})_{lower} = Iri(SQC_{oc}) - 1.96\sigma_{SQC} - (41)$

The confidence limits are given in Table 17.

Minimum Requirements to Compute SQC

It has been demonstrated that the computation of sediment quality criteria for a particular chemical requires key parameter values as well as evidence that EqP is applicable for a particular chemical. Minimum requirements for these parameters are warranted so that they provide the level of protection intended by SQC and that are within the limits of uncertainty set forth in this document. This section outlines minimum data requirements and guidance for deriving them. This is a necessary step to develop reliable parameters to be used in computing SQC. The minimum requirements for an EqP based SC are as follows.

Octanol-Water Partition Coefficient (Kow)

- Final chronic value (FCV)
- Sediment Toxicity Tests

Procedures to ensure that these data meet assumptions of the EqP approach will also be addressed.

Laboratory octanol-water partition coefficient. Kow data developed by the slow stir flask measurement technique is required. This method has been shown to provide the least amount of variability and

Table 17.—Sediment quality criteria confidence limit	s for five chemicals.
--	-----------------------

		SEDIMENT QUALIT	Y CRITERIA 95% CONFIDENCE 1	UMITS (US/Soc)
CHEMICAL	TYPE OF WATER BODY	SQCat HE/Eas		UPPER LIMIT
	Fresh Water	132	61.5	283
ACENAPTHENE	Salt Water	232	103	498
	Fresh Water	616	290	1,300
FLUORANTHENE	Sait Water	296	140	640
	Fresh Water	182	85	391
PHENANTHRENE	Salt Water	238	111	511
	Fresh Water	4.22	1.98	9.06
ENDRIN	Salt Water	0.76	0.354	1 63
	Fresh Water	11.1	5.17	23.8
DIELDAIN	Saft Water	20.4	9.50	43 6

the least bias when comparisons of Kow estimation techniques were done for Kows derived for the five chemicals for which SQC have been developed. A minimum of three Kow values are required. These values may be taken from the literature provided that methods followed yield a degree of confidence similar to that provided by the methodology used by EPA to derive Kow values.

If slow stirk flask K_{ow} values do not exist then laboratory experiments must be conducted. Measurements of K_{ow} done at the EPA ERL, Athens, Georgia were presented. At a minimum, these procedures are recommended. EPA laboratory procedures include a quality assurance and control plan. The plan includes testing the compound by spectroscopic means to ensure its identity and purity as well as running concurrent K_{ow} measurements of reference compounds which have K_{ows} that have been verified.

Final chronic value. The FCV is computed as part of the derivation of the water quality criteria for a compound, and is defined as the quotient of the Final Acute Value (FAV) and the Final Acute-Chronic Ratio (8). The data required to compute the WQC FCV are water-only toxicity tests for a variety of organisms meeting minimum data base requirements. The FCV computation and minimum database requirements are presented in the EPA document which describes methods to be used in deriving national ambient WQC (8).

WQC are based on an assessment of a compound's acute and chronic toxicity for organisms representing a range of sensitivities, most importantly most sensitive organisms. This is appropriate since the objective of WQC is to set limits based on the best estimate of organism sensitivity. The toxicity data base should therefore include all available data that meets requirements. That is, a complete search, retrieval and review for any applicable data must be conducted, to locate all preexisting toxicity data. For some compounds a WQC FCV may exist which would provide a significant amount of toxicity data. Literature searches are recommended to locate other sources of toxicity data.

A reevaluation of an already existing FCV is warranted because data post dating publication of the national FCV can be incorporated into the FCV value. Also minimum database requirements have changed since some WQC have been published. For those compounds for which WQC FCVs do not exist, compiled toxicity data are evaluated to see if minimum data requirements as put forth by EPA (8) are met. If so an FCV could then be computed. If there is not enough water only toxicity data to compute an FCV additional water only tests will be conducted so that there is enough data to satisfy minimum database requirements. Sediment toxicity test. Verification of applicability of EqP theory is required for each compound. Sediment toxicity tests can be used for this. These tests provide a sediment based LCso. Comparison of the EqP predicted LCso with the sediment LCso concentration is direct confirmation of the EqP approach. The validity of EqP is confirmed when the toxicity test results fall within the limits of uncertainty determined in this document.

47

Guidelines for conducting sediment toxicity tests ensure that the tests are uniform and are designed to incorporae the assumptions of EqP. These tests must represent a range of organic carbon content and include organisms that exhibit sensitivity to the chemical in question. The range of organic carbon must be no less than a factor of 3 and a factor of 10 is recommended. Organic carbon content shuld be no less than 0.2 percent. Replicated toxicity tests for at least two sediments are required. Organisms to be used in the sediment toxicity tests are benthic animals which are most sensitive to the compound in question. Guidelines on appropriare selection of benthic organisms is given in the American Society for Testing and Materials annual handbook [111].

Several studies are required as part of sediment toxicity testing. A water-only flow through test is required. Water-only tests are run for five concentrations of the compound in question and a control. The endpoint of interest is the 10-day mortality of the test species. This value will be compared to the pore water and sediment mortality from the sediment spiking tests discussed next.

Two sediment spiking tests are required. The first test is for the purpose of identifying sediment spiking concentrations so that pore water concentrations in spiked sediments bracket the LC₅₀ determined in the water-only test. In addition, this test is done to determine the time-to-equilibrium of the compound between the pore water and sediment. Sorption equilibrium, and assumption of Equ theory, is essential for valid porewater and sediment concentrations.

Three spiking treatments are recommended for this first test: low, medium and high concentration. The amount of compound to add to each treatment is calculated using the initial chemical weight, the % total organic carbon (TOC), % dry weight and total volume of spiked sediment. The results are sediment concentrations that bracket the predicted LC50s estimated from the water-only LCso (µg/L) and Koc-Samples for chemical analyses in bulk sediment and pore water are collected at various time intervals. Nominal sediment spike concentrations, measured sediment TOC and measured and EqP-predicted compound concentrations in sediments and pore waters are obtained for each sample period to establish time-to-equilibrium and to verify that spiking produces the appropriate concentrations in the pore waler.

in the second sediment spiking tests three sediments representing a range of organic carbon content are spiked to yield five estimated sediment concentrations to bracket the predicted sediment LC50. The amount of compound for spiking is based on similar computations as in the first sediment spiking experiment. Each treatment (sediment by concentration) is held for the appropriate time based on time to equilibrium established in the first spiking test. Day 0 samples are taken for sediment and pore water analyses. Then organisms are placed in replicated beakers and 10-day sediment toxicity tests with the equilibrated spiked sediments are conducted. Eight replicates for each treatment are required. Four replicates are used for day 10 sediment and pore water analyses while the remaining four replicates are used to assess organism mortality.

48

These experiments provide data to compute pore water toxic units and sediment toxic units (Equations 1 and 8). The results of these equations serve as direct comparisons of the predicted toxicity (Equation 1 and 8 numerator) to the observed toxicity (Equation 1 and 8 denominator). That is, the validity of EqP for a chemical is confirmed when the pore water and sediment toxic units fall within the limits of uncertainty determined in this document.

Analytical procedures. The purpose of these . procedures is to verify that:

- the WQC FCV applies to benthic organisms
- the Koc from the slow stir flask Kow is an accurate estimate of Kow

A test that the WQC FCV, which is applicable to the most sensitive water column organisms, is applica to the most sensitive benthic organisms is needed for each chemical. In computing SQC tor endrin, dieldrin, acenapthene, phenanthrene and fluoranthene, the Approximate Randomization test was applied. This is a statistical test to compare the WQC toxicity database to benthic organism toxicity. The methodology is presented previously. If it is found that benthic organisms exhibit similar or less sensitivity to a chemical than those organisms used to compute WQC, then the WQC FCV can be applied in computing an SQC. If benthic organisms shibit a greater sensitivity than the WQC organisms then toxicity experiments for benthic organisms are required.

A check on the laboratory K_{oc} must be done by comparing it to the K_{oc} computed from sediment toxicity tests. Pore water and sediment concentrations from the sediment toxicity test provide data necessary to compute K_{oc} . This K_{oc} is then compared to the K_{oc} from the slow stir flask K_{ow} .

Lastly, when a site's sediments are being studied + check to show that the SQC applies to the site in neused. National SQC may be under or over protective if 1) the species at the site are more or less sensitive than those included in the data set used to derive SQC or 2) the sediment quality characteristics of the site alter the bioavailability predicted by EqP and, ultimately, the predicted toxicity of the sediment bound chemical. Therefore, it is appropriare that sitespecific guidelines procedures address each of these conditions separately, as well as jointly. Methods to determine the applicability of national SQC to a site and to determine site specific SQC if needed are presented in the EPA guidelines document for deriving site specific sediment criteria [112].

Conclusion. Minimum database and analytical requirements must be set when deriving national sediment criteria. The reasons for this is twofold. First, the requirements provide that a level of protection intended by the criteria are met. Secondly, the reguirements provide that parameters used to compute the criteria satisfy assumptions underlying the EqP theory. The key required parameters are Kow using the slow stir flask measurement method, the WQC FCV and sediment toxicity tests. Procedures to verify that these values are appropriate to use in the SQC computation are also required. It must be shown that the FCV is protective of benthic organisms. Confidence in the Koc must also be established by comparing the Kow to the observed Koc in sediment toxicity tests. Individual sites may exhibit greater or lesser toxicity to a chemical than that predicted by SQC to an individual site. EPA procedures to test this as well as to compute site specific SQC are available.

Example Calculations

Equation 34 can be used to compute SQC_{oc} for a range of K_{ows} and FCVs. The results for several chemicals are shown in Figure 38 in the form of a nomograph. The diagonal lines are for constant FCVs as indicated. The abscissa is log10K_{ow}. For example, if a chemical has an FCV of 1.0 μ g/L and a log10K_{ow} of 4, so that K_{ow} = 10⁴, the log10 SQC_{oc} is approximately 1 and the SQC = 10¹ = 10.0 μ g chemical/g organic carbon.

As can be seen, the relationships between SQC_{oc} and the parameters that determine its magnitude, Kow and FCV, are essentially linear on a log-log basis. For a constant FCV, a 10-fold increase in Kow (one log unit) increases the SQC_{oc} by approximately 10-fold (one log unit) because Koc also increases approximately 10-fold. Thus, chemicals with similar FCVs will have larger SQC_{oc} if their Kows are larger.

The chemicals listed in Figure 37 have been chosen to illustrate the SQC_{oc} concentrations that result from applying the EqP method. The water quality concentrations are the FCVs (not the final residue values) computed as part of the development of SQC for acenaphthene, endrin, phenanthrene, dieldrin and fluoranthene or from draft or published EPA



Figure 37.—Log₁₀ SQC versus log₁₀ K_{ow} . The diagonal lines indicate the FCV values. The criteria are computed from Equation 34. K_{oc} is obtained from K_{ow} with Equation 11. The symbols indicate SQC_{oc} for the freshwater (filled) and saltwater (hatched) onteria for the listed chemicals. The vertical line connects symbols for the same chemical. The FCVs for methyl parathion, toxaphene, chlordane, parathion are from WQC or draft documents, see Table 4. The FCVs for acenapthene, endinn, phenanthrene, fluoranthene and dieldrin are computed as part of the development of SQC. The octanol/water partition coefficients for methyl parathion, toxaphene, chlordane, chlordane, parathion, endosulfan and chlorpynfos are the log mean of the values reported in the Log P database [75]. The K_{ow} s for acenapthene, endrin, phenanthrene and dieldnn are those measured from the slow stir flask method.

WQC documents (see Table 4) for the remaining chemicals plotted. Measurement of Kows for acenaphthene, endrin, phenanthrene, dieldrin and fluoranthene are from the slow stir flask method as previously presented. The Kows for the remaining chemicals are the log averages of the values reported in the Log P database [75]. While the SQCs for acenapthene, endrin, phenanthrene, fluoranthene and dieldrin meet the minimum database requirements presented in the previous section, SQCs for the remaining chemicals are for illustrative purposes only and should not be considered final SQC values. Final SQC, when published, should reflect the best current information for both FCV and Kow.

The FCVs that are available for nonionic orgamic insecticides range from approximately 0.01 μ g/L to 0.3 μ g/L, a factor of 30. The SQC_{ocs} range from approximately 0.01 μ g/g organic carbon to in excess of 10 μ g/g organic carbon, a factor of over 1,000. This increased range in values occurs because the K_{ows} of these chemicals span over two orders of magnitude. The most stringent SQC_{oc} in this example is for chlordane, a chemical with one of the lowest K_{ows} among the chemicals with an FCV of approximately 0.01 µg/L.

By contrast, the PAHs included in this example have a range of FCVs and Kows of approximately an order of magnitude. But these values vary inversely: The chemical with the larger FCV has a smaller Kow. The result is that the SQCocs are approximately the same, 240 μ g/g organic carbon. Classes of chemicals for which the effects concentrations decrease logarithmically with increasing Kows, for example, chemicals that are narcotics [113], will have SQC that are more nearly constant.

Field Data

Information on actual levels of the criteria chemicals in the environment was assembled in order to provide an indication of the relationship between the 1日本 184 小田市大学である。 二、市、市大学学生学生

SQC concentrations and the actual concentration levels observed in the sediments of U.S. surface water bodies. Three separate databases were examined:

EPA's STORET database [114],

50

- NOAA's National Status and Trends database, which focused on water bodies in coastal areas [47], and the
- Corps of Engineers database for San Francisco Bay [116].

The data that were retrieved have been summarized on probability graphs that are presented in the subsections that follow for each of the data sources. A large proportion of the observations are below detection limit values and indicate only that the actual concentration is unknown, but less than the concentration plotted. These data are plotted with a "less than" symbol. As a result, the probability plots should not be interpreted as representations of the actual probability distribution of the monitored samples. They do, however, provide a useful visual indication of the range of concentration levels of the study chemicals in natural sediments.

A suggestion of the probable extent to which problem sediments might be encountered is provided by the plot overlay showing the SQC concentration developed by this research. In the case of the STORET data, the SQC is shown as a band because the f_{oc} is not reported. The lines represent the SQC for between an $f_{oc} = 1$ to 10 percent. The other two data bases provide the necessary information on sediment organic carbon levels, and the results have been properly normalized.

Some salient features of the available field data displayed by the plots are summarized in Table 18. The SQC concentration is listed for each of the five criteria chemicals, together with the number of samples and the approximate percentage of the samples that exceed the SQC. The table also lists the sediment concentrations that are exceeded by 10, 5, and 1 percent of the measurements.

We recognize that the tabulated information represents only approximate estimates, because of the presence of large numbers of denation limit values. Nevertheless, it provides what v consider to be a reliable expectation that only a small percentage of sediment sites in the databases, less than 5 percent, will have concentrations that exceed the SQC levels.

We did not attempt a more : Jrous analysis to provide a more definitive characterization of the spatial and temporal features of the database. Some of the recorded data clearly represent multiple samples at a particular site. The very high observed concentrations are relatively few in absolute number and may reflect multiple samples at one or a few particularly contaminated sites. Some of the probability plots also show a discontinuity at the high end. This is particularly true of the Corps of Engineers data that pool results from a limited number of stations in San Francisco Bay. Until a more detailed analysis is performed, the results of the preliminary screening should be considered approximate, upper bound estimates of the probable prevalence of sediment sites that may exceed the SQC.

STORET data. A STORET data retrieval was performed to obtain a preliminary assessment of the concentrations of the criteria chemicals in the sediments of the nation's waterbodies. The data retrieved was restricted to samples measured in the period 1986 to 1990. The selection of this recent period climinated much of the older data with the higher detection limits to provide a more accurate indication of current conditions. Log probability plots concentrations are shown in Figures.38 and 39. Concentrations are shown on a dry weight basis, because s- .:ment organic carbon is not reported. The SQCs are computed on the basis of a sediment organic carbon content (foc) of 1 percent and 10 percent, which is the typical range for inland sediments. The STORET data distinguishes between the type of waterbody, and separate displays are provided for stations on streams, lakes, and estuaries.

The PAH data are shown in Figure 38. The total number of samples, and the number of detected samples are indicated on the figures. The plotted points are restricted to a subset of the total number of samples, so that the plots are legible. A few samples with detected concentrations, the solid symbols, exceed the SQC for for = 1 percent, and fewer exceed the SQC for foc = 10 percent. The nondetected data, plotted at the detection limit with "<", are below the value indicated on the plot. In fact with nundetected data included in the probability plot, the actual plotting positions of the detected data is uncertain, since the nondetected data may in fact occupy plotting positions further to the left, at lower probabilities. Thus the exceedence probabilities for the detected data are at least as large as is indicated on the plots. Approximately 5 percent or less of the detected samples exceed the foc = 1 percent SQC.

The data for endrin and dieldrin are shown in Figure 39. Similar results are obtained. Less than 3 percent of the detected dieldrin and endrin samples exceed the lower SQC.

National Status and Trends Program data. NOAA's National Status and Trends Program developed a database on the quality of marine sediments focusing on estuarine and coastal sites that are not in close proximity to known sources of contamination [116]. Figure 40 displays the distribution of sediment concentrations from the National Status and Trends Program sites for four of the five criteria chemicals

<u> </u>		FLUORANTHERE	PHENANTHRENE	/CENAPTHENE	ENDRIN	DIELORIN
BALT WATER	SQC (HE/g oc)	296	238	232	0.744	20.40
OORPS OF	NO. OF SAMPLES	231	231	130	260	282
ENGINEERS	% that exceed SQC	3%	4%	5%	0%	0%
	10% axceed - conoc	40	60	30	60	40
	5% exceed - conoc	300	300	200	150	80
	1% exceed - conc	40,000	25.000	3.000	700	600
NOAA	NO. OF SAMPLES	797	736	245	-	408
HATIOHAL STATUS	N THAT EXCEED SQC	0.2%	0.1%	0%	- 1	0%
AND TRENDS	10% exceed - conc	4	25	4		0.3
PROGRAM	PROGRAM 5% exceed - conc	7	40	7	•	0.5
	1% exceed - conc ·	40	90	40		1.0
EPA STORET ESTUARIES	SQC (ug/g) for sediment OC: 1-10%	3.0-30	2.4-24	2.3-23	.007-0.07	0.02-0.
	NO. OF SAMPLES	88	87	74	150	160
	% THAT EXCEED SQC	1%	< 1%	0%	< 10%	0%
	10% exceed conc	4	0.1	0.3	all	157/160
	5% exceed - conc	7	0.8	0.3	data are	dala are
	1% exceed - conc	40	1.0	0.5	< DL	< DL
FRESH WATER	SQC (µg/g) for sediment OC: 1-10%	6.2-82	1.6-18	1.3-13	0.04-0.42	0.1-1
EPA STORET	NO. OF SAMPLES	786	584	681	2677	3075
STREAMS	N THAT EXCEED BOO	2%	7%	< 4%	1%	< 3%
	10% exceed - conc	4	1	1	0.01	0.005
	5% exceed - conc	7	2	1.5	0.01	0.01
	1% axcesd - conc	40	?	40	0.04	0.05
PA STORET LAKES	NO. OF SAMPLES	57	50	58	478	457
	* THAT EXCEED SQC	5%	< 0%	2%	2%	< 1%
	10% exceed - conc	4	1 (?)	0.7	0.001	0.005
	5 exceed - conc	7	2 (?)	1	0.005	0.01
	1% escend - conc	40	10 (?)	3	0.100	0.08

(endrin concentrations were not measured). Sediment organic carbon concentrations were measured in this program, and ranged from less than 0.20 to 16.2 percent. The availability of for permits the plots to display both observed concentrations and the SQC value using organic carbon normalization.

Results are displayed in the plots for all samples (open symbols) and for the subset that contained organic carbon fractions greater than 0.2 percent, the limit of applicability for the EqP derived SQC (filled symbols). The EqP derived SQC is applicable to the 75 to 85 percent of all sediment samples that have focs greater than 0.2 percent. However, there is only a nominal effect on estimates of the percentage of the samples that exceed the SQC, depending on whether the full set, or the subset of samples is considered. Approximately one half a percent of subset samples exceed the phenanthrene and fluoranthene SQC, whereas none of the other SQC are exceeded. The fractions are not greatly increased if the SQC are applied to all samples.

Corps of Engineers data. A set of data from the U.S. Army Corps of Engineers monitoring program for a number of locations in various parts of San Francisco Bay has been analyzed. Table 19 identifies the locations sampled, the number of observations at each site and the period during which the results were obtained. These data were collected to examine



46T



....

Figure 39.—Comparison of sediment quality criteria for sediments containing 1 percent and 10 percent organic carbon to the distributions of two sediment pesticide concentrations from the U.S. EPA STORET Database. Data are from 1986 to 1990. Samples above the detection limit (filled symbols) and samples less than the detection limit (less than symbols) are shown. Data from U.S. EPA [114].

LOCATION	NO. OF BAMPLES PARS & PESTICIDES	YEARS
Port of San Francisco: Piers 27-29, 35, 38, 48, 70, 80 and 94	21	1988 and 1990
Fisherman's Wharf and Islais Creek	2	1990
Sulsun Channel	6	1991
West Richmond	11	1990
Pinole Shoal	44	1990
Corquinez Strait	10	1990
Mare Island Strait	6	1990
Richmond Harbor Channel	48	1990
Santa Fe Channel	6	1990
Outer and Inner Richmond Harbor Channel	6	1991
Port of Dakiand Tier II: Berths 20-23, 25, 26-30, 31, 35-38, 60-63 and 82-84	40	1989-1990
Port of Oskiand Outer and Inner Herbor	27	1990-1991
Treesure Island	5 composites	1990
Sen Leandro Bay	1 composite	1990
San Peblo Bey	6	1989-1990

Note: PAHs = Fluoranthene, Phonanthrene, Acenapthene; Pesticides = Dieldrin, Endrin.



Figure 40.—Comparison of sediment quality enteria to the distributions of acenaphthene, fluoranthene, dieldrin, and phenanthrene organic carbon normalized sediment concentrations from NOAA's National Status and Trends Program. Data are from 1984 to 1989, Samples with organic carbon greater than 0.2 percent (filled symbols) and samples for all organic carbo on contents (open symbols) are shown. Data from NOAA [47].

the quality of dredged sediments in order to determine their suitability for open water disposal. The database did not indicate what determinations were made concerning their acceptability for this purpose.

Investigators compared the frequency of occurrence (in individual samples, not dredge sites) with the SQC criteria developed using the EqP methodology. The major portion (93 percent) of the samples analyzed had organic carbon fractions greater than 0.2 percent, for which the SQC concentrations are applicable. The concentrations of each chemical measured in these sediments was normalized by the organic carbon content and the results are displayed below as probability plots to illustrate the frequency at which different levels are observed. Acsults are presented for the five criteria chemicals. A horizontal line at the concentration value of the SQC provides a reference that indicates the relationship between observed range of quality and the SQC for each chemical.

PAH results are summarized in Figure 41. Less than 5 percent of the individual samples contained concentrations in excess of the sediment quality. It is informative to note that the small set of very high concentrations are nearly all from one sample site (Treasure Island). These samples are responsible for the discontinuous pattern of the frequency distribution.

Figure 42 presents the monitoring program results for the two pesticides in the same format. In this case, virtually all of the samples were less than the varying detection limits of the analytical tests. Each of the samples for which actual measurements were obtained were at least an order of magnitude lower than the SQC. An estimate of the possible frequency distribution of sediment concentrations of dieldrin and endrin was developed by the application of an analysis technique that accounts for the varying detection limits and the presence of nondetected observations [117]. The results are illustrated by the straight line, which suggests that no appreciable number of exceedences is expected. However, the virtual absence of detected concentrations makes the



Figure 41.—Comparison of sediment quality criteria to the distributions of acenapthene, fluoranthene, and phenanthrene organic carbon normalized sediment concentrations from the U.S. Army Corps of Engineers monitoring program of San Francisco Bay. Samples with organic carbon greater than 0.2 percent (filled symbols) and samples for all organic carbon contents (open symbols) are shown. See Table 19 for description of location, number of samples, and sample period. Data from U.S. Army COE [115].

distribution estimates unreliable. They are presented only to suggest the probable relationship between the levels of these two pesticides in relation to sediment quality criteria.

CONCLUSIONS

The technical basis and data that support the use of the EqP method to generate SQC have been presented for nonionic organic chemicals. The use of organic carbon normalization is equivalent to using pore water normalization as a means of accounting for varying bioavailability (Figs. 2, 3, 5-8, 26-28). The variation in organism body burden across sediments can also be significantly reduced if organic carbon and lipid normalization are used (Figs. 29-31). For contaminated sediments, particle size effects are removed if organic carbon-normalized concentrations are compared (Figs. 17, 19, 21). The reason is that organic carbon is the proper normalization for partitioning between free dissolved chemical and sediment-bound chemical (Fig. 11).

Using pore water normalization for highly hydrophobic chemicals is complicated by chemical complexing to DOC (Fig. 13). Partitioning between



Figure 42.—Comparison of sediment quality criteria to the distributions of endrin and dieldrin organic carbon normalized sediment concentrations from the U.S. Army Corps of Engineers monitoring program of San Francisco Bay. See Table 19 for description of location, number of samples and sample period. Samples with organic carbon greater than 0.2 percent (filled symbols) and samples less than the detection limit (less than symbols) are shown. Also shown is an estimate of the distribution developed by accounting for nondetected observations (solid line). Data from U.S. Army COE [116].

pore water and sediment organic carbon from fieldcollected sediments can be rationalized if DOC complexing is taken into account (Figs. 22 and 23). However, the complexed chemical appears not to be bioavailable (Fig. 15).

These observations are consistent with the EqP model, which assumes the equivalence of wateronly exposure and the exposure from pore water and/or sediment organic carbon. Sediment quality criteria are based on organic carbon normalization because pore water normalization is complicated by DOC complexing for highly hydrophobic chemicals.

The justification for using the FCV from the WQC to define the effects level for benthic organisms has also been discussed. Water column and benthic organisms appear to have similar sensitivities for both the most sensitive species tested (Fig. 32) and all tested species (Fig. 34), Benthic colonization experi-

ments also demonstrate that WQC can be used to predict effects concentrations for benthic organisms. A direct statistical test of the equality of the distributions can be used to confirm or refute this assumption for individual chemicals (Fig. 36).

Equilibrium partitioning cannot remove all of the observed variation from sediment to sediment. It does reduce the much larger sediment-to-sediment variation that exists if no corrections for bioavailability are made (Figs. 5-8). A variation factor of approximately two to three remains (Figs. 2 and 3), which includes measurement and other sources of variability. This is not unexpected as EqP is an idealization of the actual situation. Other factors that are not considered in the model play roles in determining biological effects. Hence, it is recognized that a quantification of the uncertainty should accompany the SQC that reflect these additional sources of variation.

Research Needs

The final validation of SQC will come from field studies that are designed to evaluate the extent to which biological effects can be predicted from SQC. The colonization experiments (Table 6) are a laboratory simulation of a field validation. Sediment quality criteria can possibly be validated more easily than WQC because determining the organism exposure is more straightforward. The benthic population exposure is quantified by the organic carbon-normalized sediment concentration.

It has been suggested that the kinetics of PAH desorption from sediments control the chemical body burden of a benthic amphipod [118]. The extent to which kinetics can be important in field situations is unknown at present, and field studies would be an important component in examining this question. In addition, more laboratory sediment toxicity tests, particularly chronic tests involving multiple sediments, would also be helpful. In a typical practical application of SQC, mixtures of chemicals are involved. The extension of EqP methodology to mixtures would be of great practical value. Initial experiments indicate that it should be possible [119].

The EqP method is presently restricted to computing effects-based criteria for the protection of benthic organisms. The direct extension of this methodology for computing sediment criteria that are protective of human health, wildlife, and marketability of fish and shellfish requires that the equilibrium assumption be extended to the water column and to water column organisms. This assumption is, in general, untenable. Water column concentrations can be much lower than pore water concentrations if sulficient dilution flow is present. Conversely, upper-trophic-level organisms are at concentrations well above equilibrium values [120]. Hence, the application of the final residue values from the WQC for the computation of SQC, as was done for certain interim criteria [121], is not technically justifiable. At present, organism lipid-to-sediment organic carbon ratios, that is, BSFs (Eqn. 29), might be useful in estimating the concentration of contaminants in benthic species, for which the assumption of equilibrium is reasonable. However, a site-specific investigation (e.g., Connolly [122]) appears to be the only available method for performing an evaluation of the effect of contaminated sediments on the body burdens of upper-trophic-level organisms.

REFERENCES

 Dickson, K.L., A.W. Maki, and W.A. Brungs. eds. 1987. Fate and Effects of Sediment-Bound Chemicals in Aquatic Systems. Proceedings of Sixth Pellston Workshop, Pergamon Press, Elmsford, NY.

- Pavlou, S.P., and D.P. Weston. 1983. Initial evaluation of alternatives for development of sediment related criteria for toxic contaminants in marine waters, Puget Sound, Phase I. Development of conceptual framework. Technical Report. JRB Associates, Bellevue, WA.
- Chapman, G.A. 1987. Establishing Sediment Criteria for Chemicals - Regulatory Perspective. Pages 355-77 in K.L. Dickson, A.W. Maki, and W.A. Brungs, eds., Fate and Effects of Sediment-Bound Chemicals in Aquatic Systems. Pergamon Press, Elmsford, NY.
- Bolton, S.H., R.J. Breteler, B.W. Vigon, J.A. Scanlon, and S.L. Clark. 1985. National Perspective On Sediment Quality. EPA 68-01-6986. U.S. Environmental Protection Agency, Office of Water Regulations and Standards, Criteria and Standards Division, Washington, D.C.
- Paviou, S.P., and D.P. Weston. 1983. Initial evaluation of alternatives for development of sediment related criteria for toxic contaminants in marine waters (Puget Sound) - Phase I and Phase II. EPA 68-01-6386. U.S. Enwironmental Protection Agency, Office of Water Regulations and Standards, Criteria and Standards Division. Washington, DC.
- 6. JRB Associates. 1984. Background and review document on the development of sediment criteria. EPA 68-01-6388. U.S. Environmental Protection Agency, Office of Water Regulations and Standards, Criteria and Standards Division, Washington, DC.
- 7. Battelle. 1984. Sediment quality criteria development workshop. U.S. Environmental Protection Agency, Office of Water Regulations and Standards, Criteria and Standards Division, Washington, DC,
- Stephan, C.E., D.J. Mount, D.J. Hansen, J.H. Gentile, G.A. Chapman, and W.A. Brungs. 1985. Guidelines for deriving numerical national water quality criteria for the protection of aquatic organisms and their uses. PB85-227019. National Technical Information Service.
- Spehar, R.L., and A.R. Carlson. 1984. Derivation of sitespecific water quality criteria for cadmium and the St. Louis River basin, Duluth, Minnesota. Environ. Toxicol. Chem. 3:651-55.
- Carlson, A.R., H. Nelson, and D. Hammermeister. 1986. Development and validation of site-specific water quality criteria for copper. Environ. Toxicol. Chem. 5:997-1012.
- Luoma, S.N., and W. Bryan. 1981. A statistical assessment of the form of trace metals in oxidized sediments employing chemical extractants. Sci. Total Environ. 17:165-96.
- Chapman, P.M., and E.R. Long. 1983. The use of bioassays as part of a comprehensive approach to manne pollution assessment. Mar. Pollut. Bull. 14:81-84.
- 13. Long, E.R., and P.M. Chapman. 1985. A sediment quality triad: Measures of sediment contamination, toxicity and infaunal community composition in Puget Sound. Mar. Pollut. Bull. 16:405-15.

- 14. Barrick, R.C., D.S. Becker, D.P. Weston, and T.C. Ginn. 1985, Commencement Bay nearshore/tideflats remedial investigation. Final Report. EPA 910/9-85-134b. Prepared by Tetra Tech, for the Washington Department of Ecology and the U.S. Environmental Protection Agency. Washington, DC.
- Long, E.R., and L.G. Morgan. 1990. The potential for biological effects of sediment -- sorbed contaminants tested in the National Status and Trends Program. NOAA Technical Memorandum NOS OMA 52. Office of Oceanography and Marine Assessment, Seattle, WA.
- 16. Cowan, C.E., and C.S. Zarba. June 1987. Regulatory applications of sediment quality criteria Final Report. EPA 68-01-6986. U.S. Environmental Protection Agency, Office of Water Regulations and Standards, Criteria and Standards Division, Washington, DC.
- 17. Adams, W.J., R.A. Kimerle, and R.G. Mocher. 1985. Aquatic safety assessment of chemicals sorbed to sediments. Pages 429-53 in R.D. Cardwell, R. Purdy, and R.C. Bahner, eds., Aquatic Toxicology and Hazard Assessment: Seventh Symposium. STP 854. American Soclety for Testing and Materials, Philadelphia, PA.
- 18. Ziegeníuss, P.S., W.J. Renaudette, and W.J. Adams. 1986. Methodology for assessing the acute toxicity of chemicals sorbed to sediments: Testing the equilibrium partitioning theory. Pages 479-93 in T.J. Poston, and R. Purdy, eds., Aquatic Toxicology and Environmental Fate. 9th Volume. STP 921. American Society for Testing and Materials, Philadelphia, PA.
- 19. Swartz, R.C., D.W. Schults, T.H. DeWitt, G.R. Ditsworth, and J.O. Lamberson. 1990. Toxicity of fluoranthene in sediment to marine amphipods: A test of the equilibrium partitioning approach to sediment quality criteria. Environ. Toxicol. Chem. 9:1071-80.
- Nebeker, A.V. G.S. Schuvtema, W.L. Griffis, J.A. Barbitta, and L.A. Carey. 1989. Effect of sediment organic carbon on survival of Hyalella aztrca exposed to DDT and endrin. Environ. Toxicol. Chem. 8:705-18.
- Schuytema, G.S., A.V. Nebeker, W.L. Griffis, and C.E. Miller. 1989. Effects of freezing on toxicity of sediments contaminated with DDT and Endrin. Environ. Toxicol. Chem. 8:883-91.
- 22. Muir, D.C.G., G.P. Rawn, B.E. Townsend, W.L. Lockhart, and R. Greenhalgh. 1985. Bioconcentration of cypermethrin, deltamethrin, fenvalerate, and permethrin by Chironomus tentans larvae in sediment and twater. Environ. Toxicol. Chem. 4:51-61.
- Adams, W.J., R.A. Kimetle, and R.G. Mosher. 1983. Aquatic safety assessment of chemicals sorbed to sediments. Special Study Report No. ES-EAG-83-1. Monsanto Company, Environmental Sciences Center, St. Louis, MO
- Adams, W.J. 1987. Bioavailability of neutral lipophilic organic chemicals contained in sediments: A review. Pages 219-44 m K.L. Dickson, A.W. Maki, and W.A. Brungs, eds., Fate and Effects of Sediment-Bound

Chemicals in Aquatic Systems. Pergamon Press, Elmsford, NY.

- Karickhoff, S.W. 1984. Organic pollutant sorption in aquatic systems. J. Hydraul. Div. ASCE 110:707-35.
- O'Connor, D.J., and J. Connolly, 1980. The effect of concentration of adsorbing solids on the partition coefficient. Water Resour. 14:1317-23.
- 27. Di Toro, D.M. 1985. A particle interaction model of reversible organic chemical sorption. Chemosphere 14:1503-38.
- Benes, P., and V. Majer. 1980. Trace Chemistry of Aqueous Solutions. Elsevier, New York, NY.
- Gschwend, P.M. and S. Wu. 1985. On the constancy of sediment-water partition coefficients of hydrophobic organic pollutants. Environ. Sci. Technol. 19:90-96.
- 30. Carter, C.W., and I.H. Suffett, 1983. Interactions between dissolved humic and fulvic acids and pollutants in aquatic environments. Pages 215:30 in R.L. Swann and A. Eschenroeder, eds., Fate of Chemicals in the Environment. ACS Symposium Series 225. American Chemical Society, Washington, DC.
- Voice, T.C., C.P. Rice, and W.J. Weber, Jr. 1983. Effect of solids concentration on the sorptive partitioning of hydrophobic pollutants in aquatic systems. Environ. Sci. Technol. 17:513-18.
- Cutl, R.L., and G.A. Keolelan. 1984. Implicit-adsorbate model for apparent anomalies with organic adsorption on natural adsorbents. Environ. Sci. Technol. 18:916-22.
- Nelson, D.M., W.R. Penrose, J.O. Karttunen, and Mehlhaff. 1985. Effects of dissolved organic carbon on the adsorption properties of plutonium in natural waters. Environ. Sci. Technol. 19:127-31.
- Karickhoff, S.W., and K.R. Morris. 1985. Sorption dynamics of hydrophobic pollutants in sediment suspensions. Environ. Toxicol. Chem. 4:469-79.
- 35. Di Toro, D.M., and L. Horzempa. 1983. Reversible and resistant component model of hexachlorobiphenyl adsorption-desorption: Resuspension and dilution. Pages 89-114 in D. Mackay, S. Paterson, S.J. Eisenreich, and M.S. Simmons, eds., Physical Behavior of PCBs in the Great Lakes. Ann Arbor Science, Ann Arbor, MI.
- 36. Di Toro, D.M., J.D. Mahony, P.R. Kirchgraber, A.L. O'Byrne, L.R. Pasquale, and D.C. Piccinili. 1986. Effects of nonreversibility, particle concentration, and tonic strength on heavy metal sorption. Environ. Sci. Technol. 20:55.
- Mcliroy, L.M., J.V. DePinto, T.C. Young, and S.C. Martin. 1986. Partitioning of heavy metals to suspended solids in the Flint River, Michigan. Environ. Toxicol. Chem. 5:609-23.
- Mackay, D., and B. Powers. 1987. Sorption of hydrophysic chemicals from water: A hypothesis for the mechanism of the particle concentration effect. Chemosphere 16:745-57.

- Eadie, B.J., N.R. Morehead, and P.F. Landrum. 1990. Three-phase partitioning of hydrophobic organic compounds in Great Lakes waters. Chemosphere 20:161-78.
- Thurman, E.M. 1985. Organic Geochemistry of Natural Waters. Martinus Nijholf/Dr. W. Junk Publ., Dordrecht, The Netherlands.
- McCarthy, J.F., and B.D. Jimenez. 1985. Reduction in bioavailability to bluegills of polycyclic aromatic hydrocarbons bound to dissolved humic material. Environ, Toxicol. Chem, 4:511-21.
- 42. Landrum, P.F., S.R. Nihart, B.J. Eadie, and L.R. Herche. 1987. Reduction in bioavailability of organic contaminants to the amphipod *Pontoporeia hoyi* by dissolved organic matter of sediment interstitial waters, Environ. Toxicol. Chem. 6:11-20.
- 43. Landrum, P.F., M.D. Reinhold, S.R. Nihart, and B.J. Eadte. 1985. Predicting the bioavailability of organic xenobiotics to *Pontoportia hoyi* in the presence of humic and fulvic materials and natural dissolved organic matter. Environ. Toxicol. Chem. 4:459-67.
- 44. Prahl, F.G. 1982. The geochemistry of polycyclic aromatic hydrocarbons in Columbia River and Washington coastal sediments. Ph.D. thesis. Washington State University, Pullman, WA.
- 45. Evans, K.M., R.A. Gill, and P.W.J. Robotham. 1990. The PAH and organic content of sediment particle size fractions. Water Air and Soil Pollut. 51:13-31.
- 46. Delbeke, K., R.J. Claude, and M. Bossicart. 1990. Organochlorides in different fractions of sediments and different planktonic compartments of the Belgian Continental Shelf and Schelt Estuary. Environmental Pollution. 66:325-49.
- National Oceanic and Atmospheric Administration. 1991. Second summary of data on chemical contaminentis in sediments from the National Status and Trends Program. NOAA Technical Memorandum NOS OMA 59. Office of Oceanography and Marine Assessment, Rockville, MD.
- Brownawell, B.J., and J.W. Farrington. 1986. Biogeochemistry of PCBs in interstilial waters of a coastal marine sediment. Geochim. Cosmochim. Acta 50:157-69.
- Brownawell, B.J., and J.W. Farrington. 1985. Partitioning of PCBs in marine sediments. Pages 97-119 in A.C. Sigleo, and A. Hattori, eds., Marine and Estuarine Geochemistry. Lewis Publishers, Chelsea, MI.
- Socha, S.B., and R. Carpenter. 1987. Factors affecting pore water hydrocarbon concentrations in Puget Sound sediments. Geochim. Cosmochim. Acta 51:1237-84.
- Oliver, B.G. 1987. Biouptake of chlorinated hydrocarbons from laboratory-spiked and field sediments by oligochaete worms. Environ. Sci. Technol. 21:785-90.
- 52. Stehly, G.R. 1991. Memorandum to W.J. Berry, April 25, 1993.

 Hoke, R., and G. Ankley 1991. Results of dieldrin sediment spiking study conducted in support of USEPA development of sediment quality criteria. Memorandum to D. Hansen and D. Di Toro. June 18, 1991.

59

- Hoke, R., and G. Ankley. 1992. Results of dieldrin sediment spiking study. Airport Pond retest, conducted in support of USEPA development of sediment quality criteria. Memorandum to D. Hansen and D.DiToro. January 27, 1992.
- 55. Hoke, R. 1992. Results of dieldrin sediment spiking study conducted in support of USEPA development of sediment quality criteria. Memorandum to D. Hansen, D. Di Toro, and G. Ankley. December 2, 1992.
- Swartz, R.C. 1991. Acenaphthene and phenanthrene files. Memorandum to David Hansen, June 26, 1991.
- 57. DeWitt, T.H., R.J. Ozretich, R.C. Swartz, J.O. Lamberson, D.W. Schults, G.R. Ditsworth, J.K.P. Jones, L. Hoselton, and L.M. Smith. 1992. The influence of organic matter quality on the toxicity and partitioning of sediment association fluoranthene. Environ. Tox. Chem. 11: 197-208.
- Chiou, C.T. 1985. Partition coefficients of organic compounds in lipid-water systems and correlations with fish bioconcentration factors. Environ. Sci. Technol. 19:57-62.
- 59. Thomann, R.V., J.P. Connolly, and T.F. Parkerton. 1991. An equilibrium model of organic chemical accumulation in aquatic foodwebs with sediment interaction. Environ. Toxicol. Chem. (in press).
- 60. McFarland, V.A. 1984. Activity-based evaluation of potential bioaccumulation for Sediments. Pages 461-67 in R.L. Montgomery, and J.W. Leach, eds., Dredging and Dredged Material Disposal. Vol. 1. American Society of Civil Engineers, New York, NY.
- 61. Lake, J.L., N. Rubinstein, and S. Pavignano. 1987. Predicting bioaccumulation: Development of a simple partitioning model for use as a screening tool for regulating ocean disposal of wastes. Pages 151-66 in K.L. Dickson, A.W. Maki, and W.A. Brungs, eds., Fate and Effects of Sediment-Bound Chemicals in Aquatic Systems. Pergamon Press, Elmsford, NY.
- 62. Rubinstein, N.I., J.L. Lake, R.J. Pruell, H. Lee II. B. Taplin, J. Heltshe, R. Bowen, and S. Pavignano. 1988. Predicting bioaccumulation of Sediment-Associated Organic Contaminants: Development of a Regulatory Tool for Dredged Material Evaluation. Technical Report D-87-1, U.S. Environmental Protection Agency (Narragansett, RI) for the U.S. Army Corps of Engineers Waterway Experiment Station, Vicksburg, MS.
- Bierman, V.J., Jr., 1990. Equilibrium and biomagnification partitioning of organic chemicals in benthic animals. Environ. Sci. Technol 24:1407-12.
- 64. Leo, A.J. 1972. Relationships between partitioning solvent systems. In R.P. Gould. ed., Biological Correlations - The Hansch Approach. Advances in Chemistry Series 114. American Chemical Society, Washington, DC.

- 65. Thomann, R.V., and J.P. Connolly. 1984. Model of PCB in the Lake Michigan lake trout food chain. Environ. Sci. Technol. 18:65-71.
- 66. Mackay, D. 1979. Finding fugacity feasible. viron. Sci. Technol. 13:1218....
- 67. Stumm, W., and J.J. Morgan. 1970. Aquatic Chemistry. Wiley-Interscience, New York, NY.
- 68. Hansen, D.J., and M.E. Tagatz. 1980. A laboratory test for assessing the impacts of substances on de loping communities of benthic estuarine organisms. Pages 40-57 in J.G. Eaton, P.R. Parrish, and A.C. Hendricks, eds. Aquatic Toxicology (Third Symposium). STP 707. American Society for Testing and Materials, Philadelphia, PA.
- 69. Tagatz, M.E. 1977. Effects of pentachlorophenol on the development of estuarine communities. J. Toxicol. Environ. Health 3:501-6.
- Tagatz, M.E., C.H. Deans, G.R. Plaia, and J.D. Pool. 1983. Impact on and recovery of experimental macrobenthic communities exposed to pentachlorophenol. Northeast Gulf Science 6:131-36.
- Hansen, D.J. 1974. Aroclor 1254: effect on composition of developing estuarine animal communities in the laboratory. Contrib. Mar. Sci. 18:19-33.
- Tagatz, M.E., N.R. Gregory, and G.R. Plaia. 1982. Effects of chlorpyrifos on field- and laboratory-developed estuarine benthic communities. J. Toxicol. Environ. Health 10:411-21.
- Tagatz, M.E., and J.M. Ivey. 1981. Effects of fenvalerate on field- and laboratory-developed estuarine benthic communities. Bull. Environ. Contam. Toxicol. 27:256-67.
- 74. Tagatz, M.E., G.R. Plaia, ar 'C.H. Deans. 1985. Effects of 1,2,4-trichlorobenzene estuarine macrobenthic communities exposed via and sediment. Ecotoxicol. Environ. Saf. 10:351-60.
- Leo, A. and C. Hansch, eds. 1986. Log(P) Database and Related Parameters. Pomona College, Claremont, CA.
- 76. Woodburn, K.B., W.J. Doucette, and A.W. Andren. 1984. Generator Column Determination of Octanol/Water Partition Coefficients for 2cted Polychlorinated Biphenyl Conceners. Environ. 50. Technol. 18:457-59.
- 77. De Bruijn, J., F. Busser, W. Seinen, and J. Hermens. 1989. Determination of octanol/water partition coefficients for hydrophobic organic chemicals with the "slow-stirring" method. Environ. Toxicol. Chem. 8:499-512.
- 78. Karickhoff, S.W., and D.S. Brown. 1979. Determination of octanol/water distribution coefficients, water solubilities, and sediment/water partition coefficients for hydrophobic organic compounds. EPA 600/4-79-032. U.S. Environmental Protection Agency, Environmental Research Laboratory, Athens, G.A.
- 79. Kollig, H.P. 1988. Criteria for evaluating the reliability of literature data on environmental processes con-

stants. Toxicol. Environ. Chem. 17:287-311. Gordon and Breach, Science Publishers, Inc., Great Britain.

- Rapaport, R.A., and S.J. Eisenriech. 1984. Chromatographic determination of octanol-water partition coefficients (Kow's) for 58 polychlorinated biphenyl congeners. Environ. Sci. Technol. 18(3):163-70.
- Ellington, J.J., and F.E. Stancil, Jr. 1988. Octanol/water partition coefficients for evaluation of hazardous waste land disposal: Selected chemicals. EPA/600/M-88/010. U.S. Environmental Protection Agency, Environmental Research Laboratory, Athens, GA.
- Eadsforth, C.V., 1986. Application of reverse-phase HPLC for the determination of partition coefficients. Pest. Sci. 17:311-25.
- Brooke, D.N., A.J. Dobbs, and N. Williams. 1986. Octanol: Water partition coefficients (P): Measurement, estimation and interpretation, particularly for chemicals with P1035. Ecotoxicol. Environ. Saf. 11:251-60.
- 84. De Kock, A.C., and D.A. Lord. 1987. A simple procedure for determining octanol-water partition coefficients using reverse phase high performance liquid chromatography (RPHPLC). Chemosphere 16(1):133-42.
- Briggs, G.G. 1981. Theoretical and experimental relationships between soil adsorption, octanol-water partution coefficients, water solubilities, bioconcentration factors, and the parachor. J. Agric. Food Chem. 29:1050-59.
- 86. Banerjee, S., S.H. Yalkowsky, and S.C. Valvani, 1980. Water solubility and octanol/water partition coefficients of organics: Limitations of the solubility-partition coefficient correlation. Environ. Sci. Technol. 14(10):1227-29.
- Haky, J.E., and A.M. Young. 1984. Evaluation of a simple HPLC correlation method for the estimation of the Octanol-water partition coefficients of organic compounds. J. Liq. Chromatogr. 7(4):675-89.
- Hansch, C., and T. Fujita. 1964. A method for the correlation of biological activity and chemical structure. J. Am. Chem. Soc. 86:1616-26.
- Bruggeman, W.A., J. Van der Steen, and O. Hutzinger. 1982. Reversed-phase thin-layer chromatography of polynuclear aromatic hydrocarbons and chlorinated biphenyls: Relationship with hydrophobicity as measured by aqueous solubility and octanol-water partition coefficient. J. Chromatogr. 238:335-46.
- Rogers, K.S., and A. Cammarata. 1969. Superdelocalizability and charge density: a correlation with partition coefficients. J. Med. Chem. 12:692-93.
- Veith, G.D., N.M. Austin, and R.T. Morris. 1979. A rapid method for estimating log P for organic chemicals. Water Res. 13:43-47.
- Miller, M.M., S. Ghodbane, S.P. Wasik, Y.B. Tewari, and D.E. Martire. 1984. Aqueous solubilities, octanol/ water

partition coefficients, and entropies of melting of chlorinated benzenes and biphenyls. J. Chem. Eng. Data 29(2):184-90.

- Means, J.C., S.G. Wood, J.J. Hasset, and W.L. Banwart. 1930. Sorption of polynuclear aromatic hydrocarbons by sediments and soils. Environ. Sci. Technol. 14(12):1524-28.
- 94. Burkhard, L.P., D.W. Kuehl, and G.D. Veith. 1985. Evaluation of reverse phase liquid chromatography/mass spectrometry for estimation of n-Octanol/water partition coefficients for organic chemicals. Chemosphere 14(10):1551-60.
- 95. Mabey, W.R., J.H. Smith, R.T. Podoll, H.L. Johnson, T. Mill, T.W. Chou, J. Gates, I. Partridge, Waight, H. Jaber, and D. Vandenberg. 1982. Aquatic fate process data for organic priority pollutants. EPA-440/4-81-014. U.S. Environmental Protection Agency, Office of Water Regulations and Standards, Washington, DC.
- Yalkowsky, S.H., S.C. Valvani, and D. Mackay. 1983. Estimation of the aqueous solubility of some aromatic compounds. Residue Rev. 85:43-55.
- Miller, M.M., S.P. Wasik, G. Huang, W. Shiu, and D. Mackay. 1985. Relationships between octanol-water partition coefficient and aqueous solubility. Environ. Sci. Technol. 19(6):522-29.
- Yalkowsky, S.H., and S.C. Valvani. 1979. Solubilities and partitioning 2: Relationships between aqueous solubilities, partition coefficients, and molecular surface areas of rigid aromatic hydrocarbons. J. Chem. Eng. Data 24(2):127-29.
- Mackay, D., A. Bobra, and W.Y. Shui. 1980. Relationships between aqueous solubility and octanol-water partition coefficients. Chemosphere 9:701-11.
- 100. Kamlet, M.J. R.M. Doherty, P.W. Carr, D. Mackay, M.H. Abraham, and R.W. Taft. 1988. Linear solvation energy relationships: Parameter estimation rules that allow accurate prediction of octanol/water partition coefficients and other solubility and toxic properties of polychlorinated biphenyls and polycyclic aromatic hydrocarbons, Environ. Sci. Technol. 22(5):504-9.
- 101. Callahan, M.A., M.W. Slimak, N.W. Gabel, I.P. May, C.F. Fowler, J.R. Freed, P. Jennings, R.L. Durfee, F.C. Whitmore, B. Maestri, W.R. Mabey, B.R. Holt, and C. Gould. 1979. Water-related environmental fate of 129 prionity pollutants. Volume 2: Halogenated aliphatic hydrocarbons, halogenated ethers, monocylic aromatics, phthalate esters, polycyclic aromatic hydrocarbons, nitrosamines, and miscellaneous compcunds. EPA-r 440/4-79-029b. U.S. Environmental Protection Agency, Olice of Water Planning and Standards. Office of Water and Waste Management, Washington, DC.
- 102. Arbuckle, W.B. 1983. Estimating activity coefficients for use in calculating environmental parameters. Environ. Sci. Technol. 17(9):537-42.
- 103. Doucette, W.J., and A.W. Andren. 1987. Correlation of octanel/water partition coefficients and total molecu-

lar surface area for highly hydrophobic aromatic compounds. Environ. Sci. Technol. 21(8):821-24.

- 104. D'Amboise, M., and T. Hanai. 1982. Hydrophobicity and retention in reversed phase liquid chromatography. J. Liq. Chromatogr. 5(2):229-44.
- 105. Lyman, W.J., W.F. Rheel, and D.H. Rosenblatt. 1982. Table 1-4 in Handbook of Chemical Property Estimation Methods: Environmental Behavior of Organic Compounds. McGraw-Hill Inc., New York, NY.
- 106. Karickhoff, S.W., L.A. Carreira, C. Melton, V.K. McDaniel, A.N. Vellino, and D.E. Nute. 1989. Computer prediction of chemical reactivity — The ultimate SAR. EPA/600/M-89/017. U.S. Environmental Protection Agency, Environmental Research Laboratory, Athens, GA.
- 107. Anderson, E., G.D. Veith, and D. Weininger. 1987. SMILES: a line notation and computerized interpreter for chemical structures. EPA/600/M-87-021. U.S. Environmental Protection Agency, Duluth, MN.
- Weininger, D. 1988. SMILES, a chemical language and information system. 1. Introduction to methodology and encoding rules J. Chem. Inf. Comput. Sci. 28:31-36.
- 109. Chou, J.T., and P.C. Jurs. 1979. Computer-assisted computation of partition coefficients from molecular structures using fragment constants. J. Chem. Inf. Comput. Sci. 19(3):172-78.
- Noreen, E.W. 1989. Computer Intensive Methods for Testing Hypotheses: An Introduction. John Wiley & Sons, Inc., New York, NY.
- American Society for Testing and Materials, 1992. "Annual Book of ASTM Standards".
- 112. U.S. Environmental Protection Agency. 1993. Guidelines for the derivation of Site-Specific Sediment Quality Criteria. (In Review).
- 113. Abernethy, S., and D. Mackay. 1987. A discussion of correlations for narcosis in aquatic species. Page 1-16 in K.L. Kaiser, ed. QSAR in Environmental Toxicology II. D. Reidel Publishing Co., Dordrecht, The Netherlands.
- 114. U.S. Environmental Protection Agency. 1989. Handbook: Water Quality Control Information Systems, STORET. Washington, DC.
- 115. U.S. Army Corps of Engineers. 1991. Monitoring Program for San Francisco Bay Sediments. 1988 to 1990.
- 116. O'Connot, T.P. 1991. Concentrations of Organic Contaminants in Mollusks and Sediments at NOAA National Status and Trend Sites in the Coastal and Estuarine United States. Environ. Health Perspectives 90:69-73.
- 117. El-Shaarawi, A.H., and D.M. Dolan. 1989. Maximum Likelihood Estimation of Water Quality Concentrations from Censored Data. Can. J. Fish. Aquat. Sc. 46: pp. 1033-39.

118. Landrum, 1989. Bioavailability A toxicokinetics of polycyc, aromatic hydrocarbe orbed sediments for the amphipod *Pontororeu anyi*, Env: n. Sci. Technol. 23:588-95.

62

- 120. Connolly, J.P., and C.J. P derson 1988. A thermodynamic-based evaluation - arganic chemical accumulation in aquatic organisms. Environ. Sci. Technol. 22:99-103.
- 121. Cowan, D.F. and D.M. Di Toro. 1988. Interim sediment criteria values for nonpoplar hydrophobic compounds.

U.S. Environmental Protection Agency, Office of Water Regulations and Standards, Criteria and Standards Division, Washington, DC.

122. Connolly, J.P. 1991. Application of a food chain model to polychlorinated biphenyl contamination of the Lobster and Winter Flounder food chains in New Bedford Harbor. Environ. Sci. Technol. 25(4):760-70.

123. Hamilton, M.A., R.C. Russo, and R.V. Thurston. 1977. Trimmed Spearman-Karber method of estimating median lethal co-rentrations in toxicity bioassays. Environ. Sci. Techr. 11:714-19. Correction 12:417 (1978).